DOCUMENT DE TREBALL XREAP2013-07

Returns to Foreign Language Skills in a Developing Country: The Case of Turkey

Di Paolo, A. (AQR-IREA, XREAP) Tansel, A.



Returns to Foreign Language Skills in a Developing Country: The Case of Turkey*

Antonio Di Paolo

AQR-IREA, Department of Econometrics, University of Barcelona

08034, Barcelona, Spain

e-mail: antonio.dipaolo@ub.edu

Telephone: +34 93 40 37 150

Fax: +34 93 40 21 821

And

Aysit Tansel**

Department of Economics Middle East Technical University

06800 Ankara, Turkey

Institute for the Study of Labor (IZA), Bonn, Germany

Economic Research Forum (ERF) Cairo, Egypt

e-mail: atansel@metu.edu.tr

Telephone: +90 312 210 2057

Fax: +90 312 210 7964

This version: November 3, 2013

^{*} We would like to thank Albert Saiz who kindly commented on an earlier version of this paper. Thanks are also due to Birol Aydemir the president and Enver Taştı the vice president of the Turkish Statistical Institute for their kind help in implementing this study. Any errors are our own.

^{**}Corresponding author: Department of Economics, Middle East Technical University, 06800 Ankara Turkey. E-mail: atansel@metu.edu.tr Tel: 90 321 210 2057. Fax: 90 312 210 7964.

Returns to Foreign Language Skills in a Developing Country: The Case of Turkey

Antonio Di Paolo*, Aysit Tansel**

*AQR-IREA, Universitat de Barcelona

** Middle East Technical University, IZA & ERF

Abstract:

Foreign language skills represent a form of human capital that can be rewarded in the labor market. Drawing on data from the Adult Education Survey of 2007, this is the first study estimating returns to foreign language skills in Turkey. We contribute to the literature on the economic value of language knowledge, with a special focus on a country characterized by fast economic and social development. Although English is the most widely spoken foreign language in Turkey, we initially consider the economic value of different foreign languages among the employed males aged 25 to 65. We find positive and significant returns to proficiency in English and Russian, which increase with the level of competence. Knowledge of French and German also appears to be positively rewarded in the Turkish labor market, although their economic value seems mostly linked to an increased likelihood to hold specific occupations rather than increased earnings within occupations. Focusing on English, we also explore the heterogeneity in returns to different levels of proficiency by frequency of English use at work, birth-cohort, education, occupation and rural/urban location. The results are also robust to the endogenous specification of English language skills.

Key Words: Foreign Languages, Returns to Skills, Heterogeneity, Turkey

JEL Codes: I25, J24, J31, O15, O53

1. Introduction

Foreign language skills represent a form of human capital that can be rewarded in the labor market. Several papers highlight the positive economic value of foreign language knowledge among the native populations of developed countries. Any existence of positive returns to foreign language competences is expected to be even more relevant in developing countries. Fostering widespread foreign language knowledge of the population, alongside formal schooling, might represent a stepping stone for economic development in the globalized world (Seargeant and Erling 2011). However, there are relatively few studies on this topic in the developing countries, mainly due to data limitations.

2

This paper investigates returns to foreign language skills in the Turkish labor market. Turkey provides an interesting case for several reasons. First, the labor market value of foreign languages in Turkey has not been previously investigated. Second, during past decades Turkey experienced impressive growth rates (albeit with intermittent crisis periods), increases in international trade and commerce, tourist arrivals, and foreign direct investments, all of which contributed to the country's rapid social and economic development. At the same time, the increasing internationalization of economic and Research and Development (R&D) activities, the growing relevance of foreign tourism, the growing exposure to international trade and globalization stimulated the demand for foreign languages (Fidrmuc and Fidrmuc 2009, Fidrmuc 2011, Hoon et al. 2011). Indeed, demand for foreign languages arises in order to better communicate and interact with foreign counterparts, producers, suppliers, consumers, customers and authorities with a view to get information on the functioning of the foreign markets and overcome the linguistic and cultural barriers. Therefore, foreign language skills of the Turkish labor force are very important for firms functioning in the international arena and, in general, for increasing the potential for further economic growth and development in the country. Fostering foreign language skills would be especially important for a mid-sized emerging economy like Turkey, contributing to improved national performance in the global knowledge economy. Rising demand for foreign languages, combined with the relatively scarce supply of competences in foreign languages among Turkish workers, generates the potential for important economic rewards for foreign language skills in this country. This paper's main aim is to analyze the existence and amount of this potential economic premium.

Additionally, this paper also provides several salient contributions through the novelty of the data, the reported evidence, and the methodology used in our empirical analysis. We draw on the Adult Education Survey (AES) data — collected by the Turkish Statistical Institute (TURKSTAT) in 2007 — that contains detailed information about knowledge and use of several foreign languages. Resultantly, we are able to present an analysis of returns to different foreign languages, without constraining the focus only to English, as previously done for other developing countries. In order to keep the empirical analysis tractable, we

focus on employed males between the ages 25-65. We estimate returns to foreign language knowledge, while controlling for several human capital and labor market characteristics. With the aim of accounting for the indirect link between language and earnings through occupation, we also present estimations that control for occupation fixed-effects. Moreover, parental education is also included as an additional control, which captures the effect of unobserved factors, such as cognitive and non-cognitive skills and social networks, on earnings.

To do all this, we consider the following empirical questions. What are, on average, returns to foreign language knowledge? Are there increasing returns to different levels of skills in foreign languages? Do returns differ by the frequency of foreign language use at work? Furthermore, focusing on English, we analyze the existence of heterogeneity in returns to English skills with respect to frequency of use at work, birth-cohort, education and occupation, as well as rural/urban location. Finally, we consider several alternative econometric models that account for the endogeneity of English skills, in which we also accommodate for the interval-coding of our earnings variable and the discrete structure of English skills.

The organization of this paper is as follows. Section 2 provides background about the relevance of foreign language knowledge in Turkey. Section 3 reviews and discusses selected papers from the literature on the economic value of language skills. Section 4 describes the main characteristics of the data used. Section 5 reports the empirical results. Conclusions and policy implications appear in Section 6.

2. Background

In this section we discuss the recent developments in the Turkish economy that likely relate to increasing demand for foreign languages. These developments are related to the foreign trade policy, Turkey's foreign trade partners, the growing importance of the service sector and international tourism, as well as the increasing internationalization of economic and R&D activities, among other factors. We then highlight tendencies behind the supply of foreign language competences among the population, which, although increasing, appear insufficient to meet growing demand for foreign languages in the Turkish economy.

2.1 The Demand for Competences in Foreign Languages in Turkey

Turkey is considered as a middle-income country. It is the world's 18th largest economy. The country's per-capita income, which has nearly tripled during the past decade, currently exceeds 10,000 US dollars. Since the 1990s, the Turkish economy experienced several crises. These were the adverse effects of the 1990-1991 Gulf War, the financial crisis of 1994, the combined impacts of the Russian financial crisis together with two large earthquakes in 1999, the former of which also points to the intertwined structure of the Turkish and Russian economies, and the 2001 financial crisis. The growth rate averaged 6.8% during the period 2002-2007. Finally, Turkey experienced negative effects from the 2008-2009 global crisis. Subsequently, the economy grew over 8% in 2010 and 2011 and a little more than 2% in 2012.

Several researchers such as Adak (2010), Çetinkaya and Erdoğan (2010), Kotil and Konur (2010) and Öztürk and Acaravcı (2010) suggest that the expansion of international trade appears to be one of the most important factors driving economic growth and development in Turkey over the last decade. In parallel, increasing trade openness has boosted demand for foreign language competences in the Turkish labor market, since speaking a common (foreign) language is likely to reduce transaction costs with trade

partners¹. Relevant to this increased openness, Turkey switched, at the beginning of 1980 under the guidance of IMF and the World Bank, from import substitution policies of the 1960s and the 1970s to export promotion policies, with the introduction of the structural adjustment and stabilization policies. Following this, several additional export promotion and market-based growth policies were implemented. The 1988 financial liberalization fostered both exports and imports. As a result, total trade volume, which was only 11 billion US dollars in 1980, increased to 389 billion US dollars by 2012.

Exports have increased substantially since the 1980s. Total exports were only about 3 billion US dollars in 1980 and increased to 153 billion US dollars in 2012. There was a boom in Turkish exports trade performance and competitiveness in particular after 2000 — although this slowed during the recent global crisis (Cebeci and Fernandez, 2013). In addition, Turkey became primarily an exporter of industrial products as compared to exporting mostly traditional agricultural primary products as had historically dominated. Trade openness was almost 50% in 2012 (TURKSTAT, 2013). Therefore, the country experienced an increase in trade openness as well as a significant change in the industrial composition of exports during recent decades. Furthermore, in January 1996 Turkey entered into a Customs Union with the European Union (EU), which increased competitive pressures in the domestic economy. EU countries are Turkey's main trade partners, with Germany leading amongst these. Indeed, in 2012 about 9% of Turkey's exports went to Germany. Iraq and Iran follow Germany among Turkey's export markets, each receiving about 7% of the total exports of Turkey. These in turn were followed by the UK at 5.7%, and the United Arab Emirates (UAE) and Russia each at 5.4% (TURKSTAT, 2013).

Imports also increased substantially since the 1980s. Total imports were only 8 billion US dollars in 1980 and increased to 237 billion US dollars by 2012. In 2012 Russia was

¹ Indeed, there are numerous studies and robust evidence documenting that language barriers represent an impediment to the expansion of international trade flows. Hutchinson (2005) using a gravity model shows that among non-English speaking countries there is lower trade for those whose language is more distant from English (see Isphording and Otten 2013 for more details about the use of linguistic distance measures in applied economics). Thus linguistic distance diminishes the volume of trade1. Melitz (2008) finds that direct communication in a common language is three times more effective than indirect communication in promoting trade. Ku and Zussmann 2010 and Fidrmuc and Fidrmuc (2011) estimate gravity models augmented by FLs. They suggest that significant gains can be realized by improving the linguistic skills, highlighting the role of English as lingua franca for commerce and trade.

Turkey's leading import supplier with 11% of Turkey's total imports. Germany followed with 9% of total imports. China was Turkey's third largest import partner followed in turn by the U.S. and Italy each with 6%, and Iran with 5% of total imports (TURKSTAT, 2013). Also of note, 1.7% of Turkey's total exports were to Azerbaijan, with whom Turkey shares a dialect of the Turkish language, and an additional 2.3% of Turkey's exports are directed to the Turkic republics of former Soviet Union such as Kazakhstan, Uzbekistan, Turkmenistan and Kyrgyzstan. Anecdotal evidence suggests that for the purposes of trade and investment activities in these countries, large companies use English, mid-size companies use Russian, and small companies use local languages. There are also substantial exports to various Arabic speaking counties in the Middle East, collectively total 21% of Turkey's exports. Anecdotal evidence indicates that trade with Arabic speaking countries is conducted in English.

Foreign Direct Investment (FDI) brings financial resources as well as technological and managerial know-how to recipient countries and thus contributes to their economic growth. In such activities, foreign language skills enable communication and interactions with foreign counterparts, authorities, or customers in order to convey information about the functioning of foreign markets and reduce linguistic and cultural barriers (Kogut and Harbir, 1988 and Benito and Gripsrud, 1992). FDI flows to Turkey were only 18 million US dollars in 1980 but increased to 12 billion US dollars in 2012. These flows had a peak of 22 billion US dollars in 2007. Turkey also has been a significant overseas investor, reaching t4 billion US dollars in 2012, an increase of 73% (UNCTAD, 2013). Moreover, after the 1988 financial liberalization, many Turkish entrepreneurs invested and established business connections in Russia, in the former Soviet Republics of Central Asia, and North Africa. Another indicator of the global reach of the Turkish economy is the percentage of the Turkish enterprises directly or indirectly under foreign control. In 2009 the foreign control rate was 15.4%, up from 14.1% in 2008. Germany leads with a 17.1% share of foreign controlled production, while the USA follows closely with a 14.9% share of foreign controlled production.

In parallel, foreign language skills are also strongly connected with R&D activities both in business and in the academic world, as suggested by Fidrmuc (2011). Command of foreign languages enables R&D personnel in the business sector and in academia to follow new scientific and technological developments, and to interact with international researchers and institutions. Improving competences in foreign languages would thus increase the country's research potential, leading to more innovation and other productive investments that may promote economic growth in both the short-run (Segerstrom, 2000) and the long-run (Howitt, 1999). However, Turkey's R&D expenditures and R&D personnel are low compared to their OECD peers (Özçelik and Taymaz, 2008). The share of R&D expenditure in 2009 Gross Domestic Product (GDP) was only 0.85% in Turkey, compared to 2.9% in Germany (EUROSTAT, 2012). The number of R&D personnel (per million people) was only 680 researchers in Turkey in 2007, compared to 3,521 R&D employees in Germany (World Bank, 2011). According to the Ninth Development Plan (SPO, 2006), Turkey plans to increase its R&D expenditures and personnel. Moreover, it has been argued that Turkey has not be able to attract foreign R&D investments in several key sectors. For that reason, several policies have been implemented both at the national and local levels, with the aim of increasing the country's attractiveness for foreign investors (Karabag et al. 2011). Overall, these changes are increasing the need for foreign language skills in future years, since R&D personnel will have to be proficient to perform scientific and innovative activities, as well as to attract more international R&D investments, which in turn will enhance economic growth.

Moreover, during recent decades a rapid structural transformation took place in the Turkish labor market, with declining agricultural employment and a relative increase of service sector employment, including tourism². In the earlier periods, about half of total employment was in agriculture while currently, although agriculture is still important, half of the total employed population now works in the services sector. Since the early 1980s, the growth of tourism, in particular, has been substantial. Existing evidence suggests a positive contribution of tourism to GDP growth in Turkey (see Gunduz and Hatemi 2005

² Regarding this point, Leslie et al. (2001) highlight the importance of the foreign language skills for the development of the tourism sector which contributes to both the employment and the GDP of the country, and Tucci and Wagner (2004) show the importance of FL skills in the services sector.

and Arslanturk et al 2011 among others). Foreign arrivals, only about 10 million in 2000, reached approximately 32 million by 2012—a more than 200 percent increase within 12 years. In 2012 the most foreign tourists to Turkey arrived from Germany (16% of total foreign arrivals), followed by Russia at 11%, the UK at 8%, Bulgaria at 5%, and the Netherlands and Iran each at 4% — see TURKSTAT (2013). Interestingly, during the first six months of 2013 tourist arrivals from Russia exceeded those from Germany.

Taken together, recent high growth rates, increasing trade openness and economic internationalization, the phenomenal growth of the tourism sector and other changes in the structure of the Turkish labor market, and the on-going intensification of R&D activity in the Turkish economy represent the main factors contributing to increased demand for foreign language skills in the country. Moreover, the 1999 announcement of candidacy of Turkey for full membership in the European Union (EU) and the accession negotiations to the EU since October 2005 have also increased the demand for foreign language skills. This is particularly true in the case of English, because of its role as the international *lingua franca* for commerce and trade (Ku and Zussmann 2010, Fidrmuc 2011). Still, we also expect a growing importance of competences in German, Russian, and — to a lesser extent —French.

2.2 The Supply of Skills in Foreign Languages among the Turkish Labor Force

The corollary of this demand for foreign language competences is the supply of foreign language skills in the Turkish labor market³. In Turkey, competences in foreign languages are mainly acquired at either schools or private language centers⁴, the latter of which are

³ The mother tongue of the most people in Turkey is Turkish, which is not an Indo-European language but belongs to the Altay-Uralic language family. Turkish is the only official language. However, there are many other native languages spoken in Turkey. Most notable among these are Kurdish and Arabic. Yağmur (2001) gives the distribution of 40 different, other native languages and their estimated number of speakers in Turkey. According to the most recent Turkish Demographic Health Survey (TDHS 2008), Turkish is the mother tongue of about 82 percent of the ever married women and their partners in a nationally representative sample. Kurdish is the mother tongue of the 15 percent in the same sample, while Arabic is the mother tongue of the 2.2 percent. Further, of those whose mother tongue is Kurdish, 86 percent of women and 98 percent of men also speak Turkish. Of those whose mother tongue is Arabic, 91 percent of women and 99 percent of men also speak Turkish.

⁴ Foreign language knowledge in Turkey may also be related to migration background. For example, there have been several important waves of migration of ethnic Turks (i.e. individuals with direct or indirect Turkish origins) from Bulgaria to Turkey who would know Bulgarian as well as Russian. Moreover, more recently many Turkish immigrants (first and second generations) residing in Germany returned Turkey (Aydın, 2012), and they would have some proficiency

common across the country. During the 2010-2011 academic year, 34,905 individuals, about half of which were women, completed a course in a foreign language at a private language center. 92% of students studied English and 5% percent studied German. The remaining 3% completed courses in Arabic, French, Italian, Japanese, Spanish and Russian (TURKSTAT, 2012).

Foreign language instruction in the Turkish education system has changed significantly over time. French was the common foreign language studied in schools before the 1950s. However, English has replaced French during recent decades, and is now the most widely studied foreign language at schools in Turkey. Until the 1997 educational reform⁵, foreign language instruction in public schools started at the sixth grade and continued throughout high school, with courses running three hours per week. Moreover, the so-called 'Anatolian' high schools, which are highly-selective public high schools, offer more intensive English instruction. A few of these high schools also provide intensive training in either French or German. There are also private schools at all education levels in Turkey, where the language of all instruction is a foreign language, usually English.

Finally, Arabic is taught in religious vocation high schools. Before the Educational Reform of 1997, Arabic instruction started at the sixth grade but since then starts in ninth grade at these schools. There are also Anatolian religious vocation high schools where both Arabic and English are taught intensively. At the university level, an increasing number of public universities have adopted English as medium of instruction, either only for some degrees or for the whole university. Turkish medium-of-instruction universities have elective foreign language courses, predominantly English, and there is an increasing tendency to offer at least some degrees taught entirely in English. Finally, English is usually the main language of instruction at most private universities.

in German. The number of German and Turkish people migrating from Germany to Turkey added up to 39.6 thousand people at the peak of the global crisis in 2009 and was 32.8 thousand people in 2012 (Federal Statistical Office of Germany, 2013).

⁵ In general, the Turkish Education System experienced an increase in the emphasis given to teaching of foreign languages since after the educational reform of 1997. Moreover, the 1997 reform also extended compulsory schooling from five to eight years, covering also middle school (while before 1997 only five years of primary schooling were compulsory). However, here we refer to the pre-1997 period, since our data do not cover individuals who were affected by the 1997 reform of the Turkish education system. In any case, it seems also worth mentioning that another change occurred in 2012, which established twelve years of compulsory education and a further increase of foreign language instruction at schools which is being implemented gradually.

The Turkish government has also tried to foster competences in foreign languages among public sector workers. Since the early 1990s, civil servants receive salary premiums depending on their proficiency levels in various foreign languages. A voluntary examination is administered annually to those civil servants who would like to participate. This must be re-taken every five years to maintain qualification for the salary premium, the amount of which depends on the attained proficiency level.

In spite of such efforts to increase competences in foreign languages, and other labor market skills more generally, Turkey is characterized by a significant English language deficiency, as pointed out by Koru & Akesson (2011). This comes as OECD (2012) and Tansel (2012) emphasize the need to increase the English proficiency of the Turkish labor force in order to improve employability and labor mobility in today's globalized setting. However, according to Education First (2011 and 2012) the English Proficiency Index (EPI) of Turkey was 37.66 in 2011 and 51.19 in 2012. Accordingly, in 2011 Turkey was characterized as a "very low proficiency" country, ranking second from the bottom among the 33 countries examined. Similarly, in 2012 Turkey placed 32nd from the bottom among 54 countries, and was listed as a "low proficiency" country. A similar picture is provided by data from the special 2006 Eurobarometer Survey about languages in Europe⁶ (see European Union 2006). Those data indicate that Turkey has the highest percentage of population declaring an inability to have a conversation in a language different from the mother tongue among EU 25 countries and four candidate countries (67%, compared to the EU 25 average of 44%). Thus, the presence of growing demand for foreign languages, together with the relatively scarce, albeit growing, supply of foreign language competences among Turkish workers, generates the potential for important economic rewards to foreign language skills in the country. Quantifying this economic return and finding out the foreign languages that matter most in the labor market are the main aims of this paper.

⁶ Turkey as candidate country participated in the 2006 edition of the Eurobarometer; however, she did not participate in the subsequent edition of the same survey in 2012.

3. The Economic Return to Language Skills: Selected Literature

The relevance of language competences as an economic asset has gained substantial importance in the literature during the last decades. Indeed, language proficiency is generally considered as another form of human capital since, in the same fashion as formal schooling, it is a costly asset that is embodied in the individual and is likely to be productive in the labor market (see Chiswick and Miller, 1995, 2007 and Chiswick 2008 for a general overview). Most of the literature concerns immigrants, because competences in a host country's language are fundamental for their economic and social integration. However, the same framework can be applied for explaining the positive labor market value of skills in both local and foreign languages among the native population. There are several explanations for the positive relationship between language proficiency and earnings. First, language might directly affect productivity, because fluency in the language employed in the workplace enhances efficiency in communication among coworkers, managers, buyers and sellers, etc. Second, language itself represents a mechanism for achieving more prestigious occupations that are also likely to be better remunerated (see Chiswick and Miller 2009, Quella and Rendon 2012), and workers obtain a premium if their language skills match linguistic requirements in the workplace (see Chiswick and Miller 2010, 2013, where in the second paper the authors consider the same issue from the perspective of overeducation framework). This means that a substantial part of the positive relationship between language competences and earnings is indirect, operating through the occupational channel.

Third, language competences might be remunerated also when not directly used/relevant in the workplace, since this asset represents a positive signal for other cognitive skills from the employer's perspective. Indeed, there is substantial evidence in the literature on the improved cognitive skills of those individuals who are bilingual or who have studied a foreign language. Cooper (1987) and Olsen and Brown (1992) find higher college entrance examination scores among students who have studied a foreign language

⁷ Indeed, the results presented by Aldashev et al. (2009) suggest that the positive effect of language proficiency among immigrants in Germany is completely driven by occupational selection, given that is disappears once the endogenous selection into economic sector and occupation is controlled for.

in high school in the U.S. According to Bialystok (1999) and Adesope et al. (2010) who extensively investigated bilingualism, the bilingual individuals have a generalized cognitive advantage over monolinguals in the so called executive functions involving mental flexibility, inhibitory control, attention control, and task switching as well as creativity, flexibility, and originality in problem solving (Leikin, 2012). Thus, in this context, language competences could increase earnings directly by raising an individual's productivity. They could also increase earnings indirectly, favoring the access to better remunerated occupations in which language is important, or by signaling to an employer regarding the quality of education, ability and cognitive skills, and potential productivity.

The importance of knowing the language of the host country for the immigrants has been extensively studied in the context of many countries. Such studies include, among others, Chiswick and Miller (1995) in Australia, Dustmann (1994), Dustmann and van Soest (2001 and 2002) in Germany, Berman, Lang and Siniver (2003) and Lang and Siniver (2009) in Israel, Leslie and Lindley (2001), Shields and Price (2002) and Dustmann and Fabbri (2003) in the UK, and Bleakley and Chin (2004) in the U.S. It is well established in this literature that immigrants with destination country language skills obtain a positive (overall) earnings premium. Moreover, it seems that the importance of language proficiency among immigrants goes beyond the labor market, because it also improves social integration and assimilation in the host country, as recently shown by Bleakley and Chin (2010).

There is also a second parallel strand of research that focuses on the case of non-immigrants in multilingual labor markets. Shapiro and Stelcner (1997) and Albouy (2008) consider the case of Canada, where the latter study finds earnings premium to French skills among Anglophones in Quebec. Several other developed countries characterized by multilingual realities have also been investigated (see Klein, 2003 for Luxemburg, Henley & Jones, 2005 for Wales, Grin and Sfreddo, 1998 and Cattaneo and Winkelman, 2005 for Switzerland, Rendon, 2007, Di Paolo, 2011 and Di Paolo and Raymond 2012 for Catalonia —Spain) and the results obtained are usually consistent with the hypothesis that local language skills are remunerated in the labor market.

Moreover, a growing number of papers consider the return to foreign language skills among the native population. The relationship between foreign language knowledge and labor market outcomes in developed countries has been considered in the work by Saiz and Zoido (2005), who studied the return to foreign languages using a sample of US college graduates. Willams (2011) reports significant earnings premiums for English usage at work in twelve European countries, as well as for the use of other languages, especially French and German, in some cases. Ginsburgh and Prieto-Rodriguez (2011), who also focused on several European countries, confirmed the existence of a substantial return to English proficiency. Also Lang and Siniver (2009), who analyzed the case of English in Israel (as well as Hebrew among immigrants from Russia) shows that this language knowledge is significantly remunerated in the Israeli labor market for both immigrants and natives, although the return to English skills appears heterogeneous for different groups of workers.

The economic return to English proficiency has also been analyzed in some developing countries, such as Latvia and Estonia, where Toomet (2011) found that skills in local languages are not remunerated in these countries while English proficiency produces a significant earnings premium. Levinsohn (2007) and Casale and Posel (2011) reported high returns to English competences in South Africa and Azam et al. (2013) also obtained substantial earning return to skills in English in India, especially among male workers. From this evidence, English skills definitely appear to be a valuable asset in developing countries. Our study resembles to the last group concerned with developing countries, since we investigate the return to foreign language skills in Turkey. However, it should be noticed that in both South Africa and India English is the former colonial language⁸ and currently one of the official languages, whereas this is not the case in Turkey, Latvia or Estonia. Indeed, in these countries English is a non-native and non-official language. In this sense our study is close to the paper by Toomet (2011), except that he considers the case of

-

⁸ Angrist and Lavy (1997) estimate the return to proficiency in French in Morocco, which is also the colonial language and was used as language of instruction until 1983 (and was replaced by Arabic since then). They also found that the return to education were substantially lower for those who were schooled in Arabic (relative to those who received instruction in French), mostly because of reduced French skills. A similar case has been considered by Angrist et al (2008), who studied the effect of the change of the instruction language in Porto Rico — which switched from English to Spanish in 1949. They found no effect of this language-of-instruction reform on English skills among the affected population.

Russian minority in Latvia and Estonia, whereas we consider the total native population of Turkey. In addition, unlike the previous studies concerned with developing economies, we first consider the return to several foreign languages spoken in Turkey (in a similar fashion than in Willams, 2011) and then we analyze more deeply the English language skills, given that it represents the more common foreign language in Turkey as well as in many other non-English speaking countries.

In parallel to the empirical evidence, the theory behind foreign language acquisition has been developed in a game theoretic framework, starting with the pioneer work by Selten and Pool (1991), which highlights the importance of benefits and costs of foreign language acquisition. The subsequent papers by Church and King (1993), Ginsburgh et al. (2007) and Gabszewicz et al. (2011a, 2011b) point out the relevance of network externalities in foreign language acquisition, suggesting that the incentives to learn a given non-native language would be higher the greater the size of the community that speak the language (relative to the population that speaks the individual's native language). If translated to the labor market perspective, this theoretical prediction suggests that the benefits from learning a foreign language should increase with the "labor market relevance" of that language.

4. Data and Descriptive Statistics

The empirical analysis is based on nationally-representative Turkish data from the Adult Education Survey (AES). The AES was carried out in 2007 in all European Union member states, EFTA, and candidate countries, including Turkey, with the aim of obtaining information about adult education activities and lifelong learning. This survey is especially appealing for our purposes, since it contains detailed information about foreign language knowledge, skills and use, together with socio-demographic characteristics and labor market characteristics. The overall sample includes 39,478 individuals aged 18 and over. Our main goal consists of analyzing the relationship between foreign language knowledge and labor market earnings. We restrict the sample to males aged 25-65 who were regularly

employed at the time of the survey⁹. We excluded part-time workers, because they might have a different attachment to the labor market. There are very few immigrant males in the sample (less than 2%), and are also dropped from our selected sample. After deleting the observations with missing information about earnings or other relevant variables, we ended up with a final sample of 9,194 male workers.

The AES survey contains several questions about foreign language (FL henceforth) knowledge. Individuals are asked about their knowledge of up to 7 FLs. In the case of having some knowledge of at least one FL, individuals report detailed information about the two FLs they know best. Specifically, the questionnaire asks about the level of skills of the two best known FLs, the way in which they learnt that languages, as well as their frequency of use at work and for leisure. Table 1 shows the basic descriptive statistics about the knowledge of FLs in our selected sample.

[TABLE 1 AROUND HERE]

Roughly 67% of the individuals in the sample do not speak any FLs — highlighting again the relatively scarce endowment of FL competences in Turkey¹⁰. Of the one in three individuals able to speak at least one FL, most only speak just one. FL knowledge is more common among the younger cohort, those with greater educational attainment, and among white collar workers, especially if otherwise high-skilled. Knowing at least one FL is also more common in urban areas than in rural areas¹¹.

Table 2 reports the specific languages spoken among those who declare some knowledge of FLs. It appears that English is the most widely known FL, with almost 80%

⁹ Females are excluded from the analysis in order to avoid problems of endogenous selection into labor market participation and employment. We consider individuals aged between 25 and 65 because regular schooling is usually completed before 25 years of age and can therefore be taken as exogenous, which helps to limit the potential endogeneity of schooling in the earning regressions. Selection into employment among males could also be an issue. For this reason, we controlled for endogenous selection into employment among males and the results are virtually unchanged (full results are available upon request). Therefore, we decided to focus on employed males, implying that we aim at providing evidences that are consistent for the selected sample.

evidences that are consistent for the selected sample.

10 Indeed, raw data from AES suggest that Turkey is the country with the highest percentage of individuals who declare no knowledge of any FL (75.5% in the whole sample), compared to the Europe-27 average of 37.5%. The numbers from Turkey are relatively similar only to those from Hungary (74.8%) and Romania (69.6%). More details can be consulted here: http://epp.eurostat.ec.europa.eu/portal/page/portal/education/data/database. Notice that this evidence is consistent with the results obtained in the Eurobarometer Survey of 2006, albeit slightly worse (probably because the AES data discussed above refer to individuals aged 25-65).

¹¹ Locations with population over 20,000 are defined as urban and the locations with population 20,000 or less are defined as rural areas.

of those who possess some knowledge of FLs declaring that English is one of the languages they know — at least to some extent. This evidence reflects the preeminence of English as Lingua Franca during recent decades. German represents the second most frequent language known by 12%, and considerably less common than English. The number of German speakers in Turkey reflects Germany's position both as an important trade partner for Turkey, with the largest share of Turkey' exports as noted above, and also as a traditional destination country for Turkish immigrants. Arabic is the third most frequent language (9.5%), which is taught as subject in religious vocation high schools and might be common among the indigenous population in the south-southeast of the country as well as to people with some migration experience in MENA countries (which were alternative migration destinations during the '80s), followed by French (7.3%), which was widely taught as part of the oldest generation's schooling. Less common are Russian (2.6%) and Bulgarian (0.4%) both of which are not taught in the school system. However, these two languages are likely to be commonly known by ethnic Turks who migrated from Bulgaria, as well as returning Turkish workers from the migration wave towards Russia and Central Asia that occurred in since the 1990s (Tansel and Yaşar, 2010).

[TABLE 2 AROUND HERE]

Crossing this information with birth-cohort reveals that English is relatively more common within the younger cohort, as is Russian, while the knowledge of German, Arabic and French is somewhat higher among older populations. Disentangling the frequency of FL knowledge by education suggests that, on the one hand, English is mostly learnt through the schooling process for younger cohorts while French was more commonly learnt at school among older cohorts. On the other hand, Russian, German, and especially Arabic are significantly more common among the less educated. In particular, almost 50 percent of Arabic speakers sampled have 5 or fewer years of schooling. Especially those who know German among the less educated may be return migrants from Germany, but, unfortunately, we do not have information about previous migration experiences. The most noticeable evidence obtained from separating the sample by occupation is that, as expected, English is more frequently known among white collar workers, while blue collar workers who declare to know FLs are relatively more likely to know Arabic. Finally, the incidence of Arabic knowledge and — to a lesser extent — of German knowledge appear to be

relatively higher in rural areas. In the case of German, this evidence might be reflecting previous (direct or indirect) migration background from Germany. Regarding Arabic, its incidence among low educated individuals residing in rural areas might mirror ethnic identities with Arabic roots¹².

In Table 3 we focus more deeply on the FL individuals know better. Foremost, it emerges that English represents the primary FL for about three-fourths of FL speakers, followed by German (8.4%), Arabic (6.9%) and French (5.2%). Additional evidence can be obtained by considering the information about the way in which people learnt the best FL they know (not shown here). Remarkably, although most of the people declare they acquired English skills at school (79% among those who affirm English to be the best FL they know), learning this language in a private course (10%) as well as by self-learning (8%) are relatively common options. On the contrary, 94% of French speakers learnt this language at school. The evidence about German, Russian and Bulgarian are consistent with the migration/ethnic background hypothesis, since the share of individuals who declare to have learnt these language abroad is significantly higher than for other languages. Moreover, among Bulgarian speakers, the schooling mechanism it's also common, since they might be Ethnic Turks who received some schooling in Bulgaria and then return to Turkey. Finally, albeit 29% of those that consider Arabic as the best FL they know learnt Arabic at (religious vocation) school, 45% of individuals declare they acquired the language within the family (which is in line with the idea of ethnic origins of Arabic roots).

We can also go into more detail about the quality level of FL skills¹³. Among those who declare English to be their first FL, 55% report having a basic level, about 32% have regular skills and only 13% are fully proficient in English with advanced skills. The

¹² Unfortunately, discern this point is not possible since the Turkish questionnaire of the AES survey does not include specific questions about mother tongue (which are indeed included for other countries). Therefore, the information about Arabic knowledge should be taken with caution, since its distinction with ethnic background is somewhat subtle.

¹³ Notice that the AES was in the case of the action of the AES was included for other countries.

¹³ Notice that the AES questionnaire contemplate four different self-reported levels of command of foreign languages, namely 1 "I can understand and speak a few words and sentences", 2 "I can understand and use the most general daily expressions", 3 "in the instances where the language is used in a clear fashion, I can understand the essence and express the experiences and events in a printed text" and 4 "I can understand and use the language in a flexible (fluent) manner in various subjects involving a series of difficult texts. I am almost completely competent in this language". Given the low number of cases for levels 3 and 4 we decided to group these two FL command levels into one. Therefore, in the empirical analysis we will use 3 separate levels of skills: 1) basic skills (corresponding the original level 1 in the survey), 2) regular skills (corresponding to level 2) and advanced skills (corresponding to either level 3 or 4 in the AES questionnaire).

distribution of German skills follows a similar pattern, whereas French skills are mostly concentrated into the basic level and those who claim Arabic to be their first FL are relatively more likely to have an advanced level of command of that language¹⁴.

[TABLE 3 AROUND HERE]

Finally, raw earning differentials by general FL knowledge are reported in Table 4. The AES survey includes net monthly earnings from the main job (in Turkish liras), which are reported in five distinct intervals. Tabulating interval-coded monthly earnings shows that the incidence of top-coded earnings is significantly higher among those who speak at least one FL, while the frequency of low-earnings is also lower among this sub-group of workers in our sample. This means that, to some extent, knowing a FL is generally associated with higher earnings. Similar evidence can be obtained computing average monthly earnings¹⁵, which are markedly higher among FL speakers. However, not all FLs are associated with higher earnings, as shown in the rest of the columns in Table 4.

[TABLE 4 AROUND HERE]

Indeed, the knowledge of German, English, Russian, or French is clearly associated with higher earnings—i.e. higher relative frequency in higher earnings categories and lower frequency in lower earnings categories, as well as higher average earnings. However, this is clearly not the case for Arabic and Bulgarian, which instead seem associated with lower earnings. Nevertheless, the relationship between FL knowledge and earnings that we observe in the raw data might be confounded by other individual and labor market characteristics that are likely to co-vary with both FL knowledge and earnings. Therefore, in the next section we analyze the return to FL knowledge in a regression framework, which would provide the *ceteris paribus* or conditional association between FL knowledge and skills and labor market earnings. The complete list of explanatory variables used in the empirical analysis is provided in Table 1A in the Appendix (the content of each variable is self-explanatory), together with some descriptive statistics for different sub-samples of workers.

¹⁴ The distribution of skills in Russian indicates that the majority of those who declare this language to be the first FL they know report regular skills, while Bulgarian skills are uniformly distributed across the three categories. However, these numbers should be read with caution because of the reduced number of cases in the selected sample.

¹⁵ Average monthly earnings are obtained regressing interval-coded earnings on a constant, using the "Interval Regression" method ("intreg" command in STATA) developed by Stewart (1983).

5. Empirical Results

5.1 Foreign Language Knowledge and Earnings

In this section we study the conditional relationship between FL knowledge and labor market earnings. Table 5 contains the results from several regressions of (logged) intervalcoded earnings¹⁶ on typical human capital and labor market variables, plus different indicators of FL knowledge. First, we include dummies for the number of FLs that are known by each individual in the sample. Second, we estimate several separate equations containing a dummy for each specific FL, considering English, French, German, Arabic, Bulgarian, and Russian respectively. Third, the set of indicators for general knowledge of each of these six different FLs is jointly included in the earnings regression. Finally, using this more complete specification, we add in two subsequent steps: occupation fixed-effects and dummies for parental education. The inclusion of occupation fixed-effects (two-digit ISCO88 classification) informs us about the extent to which the relationship between FL knowledge and earnings is indirect, working through the occupational channel — i.e. individuals who know FLs earn more because they are attracted into better paid occupations. Furthermore, controlling for the highest parental education among the two parents should limit the potential bias provoked by the omission of relevant unobservable characteristics, such as cognitive-and non-cognitive ability and social networks.

[TABLE 5 AROUND HERE]

The estimates of the control variables are quite standard and are just briefly discussed in what follows. The earning return to one additional year of schooling ranges between 7.4% and 8.1% when occupation is not included in the model. The noticeable evidence is that the return to schooling in specifications that contain single dummies for FL knowledge is roughly the same when no language control is included (i.e. 8.1%). However, it falls to 7.4% with the inclusion of the English dummy, suggesting that schooling, especially for younger cohorts of workers, represents an important mechanism to foster English knowledge. As expected, the return to schooling decreases after controlling for occupation

_

¹⁶ As briefly commented above, net monthly earnings are reported in intervals. Therefore, regression analysis is based on the Interval Regression model (Stewart, 1983), estimated by Maximum Likelihood ("intreg" command in STATA). Similar results are obtained employing the typical mid-point approximation (available upon request).

fixed effects, indicating that occupation mediates the conditional association between schooling and earnings. Potential experience presents the typical inverted U-shaped pattern, which is quite stable across specifications. We also include dummies for the type of employment¹⁷. Compared to employees with a permanent contract, salaried workers with a fixed-term contract earn about 24-25% less. Also self-employed workers obtain slightly lower earnings, albeit this differential disappears once controlling for occupation. On the other hand, employers receive on average 41% higher earnings than the reference group; this positive earning differential decreases to 36% when estimated within the same occupation. We also control for urban location, in order to account for the uneven structure of the labor market across the Turkish territory. This dummy shows that workers in urban areas earn 25% more on average than those in rural areas. Moreover, accounting for urban/rural residential location, slightly reduces the return to English and Russian knowledge and — even more significantly — to Arabic knowledge, capturing some part of local labor market heterogeneity.

The first column of Table 5 shows that having some knowledge of one FL is associated, on average, with 9% higher earnings. This positive return increases to 14% in the case of knowing two FLs and up to 32% in the infrequent case of having some level of command of three or more FLs. However, unpacking the return to each distinct FL confirms that not all languages are equally rewarded in the labor market. In fact, while English has a clearly significant earnings return—around 11%, the estimate for French is positive but statistically insignificant. Having some knowledge of German is positively rewarded with estimated return of about 6.4%, but knowing either Arabic or Bulgarian seems to be conditionally unrelated to labor market earnings. Finally, we obtain a noticeably high and significant return to Russian knowledge, which is associated with 19% higher monthly earnings. When we simultaneously include all the dummies for FLs knowledge, the point estimates of English and German knowledge remain virtually unchanged. We observe that the return to French knowledge becomes significant and slightly higher when all the languages are taken into account. This means that, conditional on the general command of other FLs (mainly

¹⁷ The point estimates of our coefficients of interest (i.e. those associated with FL knowledge indicators) are invariant to the inclusion of dummies for the type of employment. After testing different alternative specification, the Bayesian Information Criterion indicates that the model performs better when type of employment dummies are included.

English), mastering French could also represent a labor market asset. However, the return to Russian knowledge is somewhat reduced in this full specification, indicating that at least some part of the estimated return to Russian knowledge might be driven by other "language-related" unobservable characteristics.

After controlling for occupation fixed effects¹⁸, we obtain a lower return to English knowledge. It appears that occupation itself accounts for about 20% of the return to English knowledge. However, even among workers within the same occupation, the earnings return to general English knowledge is substantial and strongly significant. A similar pattern is observed for the case of French and German knowledge. Notice that, for the latter language, the estimated return is no longer significant when estimated within occupations, suggesting that the economic value of German knowledge in Turkey is mostly produced through the occupational channel. No significant changes are observed for Russian and Bulgarian knowledge, whilst controlling for occupation yields a slightly significant negative coefficient for Arabic¹⁹.

In the final step, we also include dummies for parental education, which might capture some effect of potentially relevant unobservables (correlated with parental education), such as cognitive-and-non-cognitive ability and social networks. Notably, parental education is a significant predictor of monthly earnings, pointing to a certain degree of social segmentation in the Turkish labor market. Moreover, we observe a very modest reduction in the point estimates of the English knowledge and imperceptible changes in the other language coefficients after including parental education.

-

¹⁸ We also tried to include dummies for economic sector. However, once controlling for occupation, the inclusion of sector fixed effects barely affected the return to FL knowledge (in a similar fashion as in Azam et al., 2013). Therefore, sector dummies have been suppressed in order to simplify the presentation (the full results are available upon request). It might be argued that the inclusion of occupation fixed effects represents "bad controls" (Angrist and Pischke, 2009), in the sense that the estimation of the treatment effect's parameter (i.e. FL return) is confounded by the inclusion of controls that depend on the treatment itself (i.e. occupation). Therefore, under positive occupational sorting, the mediating impact of occupation in the language-earning relationship is likely to represent a lower bound of the whole relevance of occupation.

occupation.

19 This result, together with the evidence that knowledge of Arabic is more common among older and less educated workers located mostly in rural areas, point out that the negative relationship between Arabic language knowledge and earnings is probably due to the fact that Arabic knowledge does not always represent an "investment" in a FL and thus this negative return should be taken with caution. We believe that disposing of information about the region of residence would help in disentangling this evidence since we expect that this negative differential is driven by residents in the southeastern part of the country. However, the regional identifiers of the Turkish AES 2007 data are not released due to data protection legislation.

This last evidence points out the robustness of our results in the light of potential omitted variables bias. Indeed, it might be argued that the positive relationship between FL knowledge and earnings is mostly driven by unobserved individual heterogeneity. However, we argue that this is unlikely to be the case. Following Lang and Siniver (2009), obtaining similar estimates when all the possible FLs are simultaneously included in the earnings equation can be taken as indicative of a barely relevant impact of unobserved ability in biasing our estimated coefficients. The idea is that the ability to learn two or more different FLs should be similarly correlated with general unobservable skills. Indeed, if knowing different languages mainly depends on unobserved ability, we should observe significant changes in the estimated coefficients when all FL dummies are simultaneously included. The opposite evidence can be taken as suggestive of a relatively limited bias due to unobserved ability. In our case, the coefficients associated to English and German knowledge are virtually unaffected by the inclusion of other FL indicators in the regression and no statistically significant differences are observed for other languages. This again supports the fact that our results are not just reflecting unobserved individual heterogeneity. A more convincing argument has been proposed by Saiz and Zoido (2005), who estimate the return to FLs among US college graduates using two-period panel data, exploiting past information about FL knowledge. They argue that if unobserved ability is the main driver of the return to language skills, one should observe similar earnings returns for those who currently speak a FL and those who were able to do it only in the past but not in the current period²⁰.

Unfortunately, in our cross-section data we only have information about current language knowledge. However, we argue that if the positive association between FL knowledge and earnings is just due to the fact that more able individuals are more likely to know at least one FL and also to earn more, we should not find any significant return to the

²⁰ Indeed, Saiz and Zoido (2005) obtain a statistically insignificant coefficient for the indicator capturing those who were able to speak a FL in the initial observation but not in the current period, suggesting that the impact of omitted ability should be rather limited. Their results using panel data and propensity score matching support this hypothesis. Also the panel data estimates for the return to English proficiency reported in Lang and Siniver (2009) are virtually identical to those obtained by OLS (albeit the return to Hebrew proficiency appear to be lower in their longitudinal estimates), which means that FL knowledge should, at least in part, represent an investment in human capital that is remunerated in the labor market rather than just reflecting unobserved ability.

knowledge of an additional language among the subsample of speakers of a (common) FL. This intuition derives from the fact that this subsample should be more homogeneous in terms of unobserved attributes that facilitate FL knowledge and probably affect earnings potential. Therefore, we perform additional language-augmented regressions using the subsample of workers that declare English to be the best FL language they know, as English is the most common FL in the sample. Selected results are shown in Table 2A in the Appendix and, with the exception of Arabic, display a positive return to the additional investment in human capital enclosed in the knowledge of other languages especially French and Russian among English speakers. In any case, the evidence suggests that the return to FL knowledge is not just a mirror of ability bias. In what follows we analyze in more detail the economic value of different levels of skills in FLs and its heterogeneity for different subgroups of workers. After that, we check for the robustness of our results with respect to the potential endogeneity of English competences in a more compelling way (see section 5.4).

5.2 Earnings Return to Different Skill Levels in FLs

The results in the previous section point out that having some knowledge of languages other than the mother-tongue generally has a market value in Turkey. However, if the labor market pays a different price for different levels of command of a language, general levels of FL knowledge might be just a partial picture of the earnings return to this human capital asset. Hence, we exploit the available information about different skill levels in the best FL an individual knows²¹. Table 6 reports the results of several earning regressions with dummies for different level of competences in each FL (columns 1-4). Finally, dummies for skills in all the relevant languages are jointly included into one single equation (column 5),

-

²¹ Albeit we also dispose of information about skills in the second FL, we just focus on the first FL because of the reduced number of individuals who know more than one FL. Indeed, the returns to skills in second FLs are mostly insignificant and the estimates for skills in the first FL are robust to the inclusion of second foreign language's skills. Moreover, we consider a more parsimonious specification that incorporates only skills in languages that have a positive return in this step of our analysis (i.e. we do not include skills in either Arabic or Bulgarian, given that these languages appear not to be rewarded in the labor market). Notice also that the estimated models contain the same set of controls reported in Table 5, whose estimates are roughly the same and are neither reported nor discussed here for brevity reasons. Further, we also tried to control for the way in which people learnt the best FL they know and the results were virtually the same.

which is successively augmented by occupation fixed effects and parental education as additional controls (columns 6-7).

[TABLE 6 AROUND HERE]

The results concerning English skills reveal that there is a positive and increasing earnings return to different levels of proficiency in English—an up to 45% increase in earnings for advanced skills. It seems worth highlighting that our estimates of the returns to different skill levels in English are very close to those reported by Azam et al. (2013) in India, who found that for men in India return to speaking fluent English is 35% and for speaking little English is 13% relative to men who speak no English²². Regarding Russian skills, the results from our estimation reveal that only very proficient individuals are able to obtain a significant remuneration for their competences in Russian of about 37%, although returns to lower skill levels are imprecisely estimated. In contrast, when dummies for French and German skills are individually included in the earnings regression, we do not find any significant return to skills in these two languages.

More compelling evidence can be obtained by including all the dummies for the skill levels in each of the four relevant FLs in the earning equation. Indeed, the return to English skills is almost unaffected by this exercise, except that basic English skills now receive a slightly higher remuneration of 4.1%. Further, there is a slightly significant return to regular French skills and to basic or advanced skills in German. Moreover, the market price of advanced Russian skills is still positive, significant and slightly higher, while that of the regular Russian skills becomes now marginally statistically significant and somewhat higher. Adding occupation fixed effects to this model produces a modest reduction in the estimated return to FL skills, indicating again that FL knowledge also affects earnings indirectly — via occupational attainments. Specifically, returns to regular French skills lose statistical significance and returns to regular Russian skills become marginally higher and more significant when estimated within the same occupation. Finally, when parental

-

²² Our results regarding the return to English skills are also in line to what reported elsewhere in the literature, for developed (Lang and Siniver, 2009, Willams, 2011 and Ginsburgh and Prieto-Rodríguez, 2011) and developing countries (Levinsohn, 2007 and Casale and Posel, 2011 Toomet, 2011), albeit that the indicators for English proficiency are not always directly comparable. The main exception are the results obtained by Saiz and Zoido (2005), who find 2-3% return to speaking a second language for college graduates in the US. This relatively low return is in all likelihood due to the fact that English represents a *lingua franca* for international trade, although it might be also a consequence of the peculiarity of the sample used. This is also consistent with the evidence reported by Willams (2011) for UK and Ireland, where no significant returns are obtained for the use of FLs (other than English) at work.

education is included as further control, we observe an additional reduction in the estimated return to FL skills, which is completely imperceptible from a statistical point of view.

Our analysis reveals that competences in FLs are positively rewarded in the Turkish labor market, although not all the languages have the same return in terms of earnings. In fact, there is no earning premium for knowing Bulgarian or Arabic. There is some evidence of positive return to French and German knowledge, although the economic value of competences in these two languages appears to be conditional on specific occupational attainments and on the knowledge of other FLs (i.e. English). Competences in Russian are more clearly associated with higher earnings, especially in the case of having advanced skills. Last but not least, returns to English knowledge are clearly positive and statistically robust in several specifications (i.e. controlling for other languages, occupation and parental education). Moreover, the earnings premium of English knowledge increases with proficiency in the language, highlighting the similarity between FL knowledge and other forms of human capital. Given this, in the rest of the paper we focus more deeply on the economic value of English competence. We do so also considering that English represents the most widely spoken FL in Turkey — as well as in other non-English speaking countries in Europe (see European Union 2006, 2012) — and is commonly used as the *lingua franca* for commerce and trade (see Ku & Zussmann 2010, Fidrmuc 2011). However, so far we considered returns to different levels of competences in English to be the same for all the Turkish male workers, although there are several reasons, including existing evidence, to consider the existence of heterogeneity in the return to English skills, which is the subject of the next section.

5.3 Heterogeneous Returns to English Skills

5.3.1 Returns to English Skills by Frequency of English Use at Work

Following the previous literature on the return to FL skills (Saiz and Zoido, 2005, Lang and Siniver, 2009, Casale and Posel, 2011 and Azam et al., 2013 among others), we consider the possibility of heterogeneous returns to English skills according to several available observed characteristics. First, as noticed by Grin et al. (2010), until now we implicitly consider that skills in English are remunerated because they are used in the labor market. If this is true, the return to English competences should increase — at least to a certain extent — with the degree to which English is used at work. However, it might also be the case that English proficiency constitutes a signal for other valuable skills from the employer's perspective. This means that being proficient in English would be remunerated even if not actually used at work. We use the information about frequency of English use at work to check for the potential heterogeneity of return to English depending on the frequency of its use in the workplace. Table 7 reports selected coefficients from different equations allowing the returns to English skills to be different according to the frequency of its use at work²³ (model 1), which are also estimated controlling for occupation fixed effects (model 2) and for parental education (model 3).

The results suggest a concave relationship between returns to English skills and the rate at which it is claimed to be used in the workplace. Actually having regular skills in English receives better remuneration if this language is used at least once per month, since the premium decreases in the case of more recurrent use of English at work. The evidence for advanced skills is similar, except that the premium decreases only in the case of daily use. Moreover, the shape of this concavity is somewhat more pronounced when estimated within the same occupation, implying that English competence serves as a signal for acceding to certain jobs. In fact, there is a positive remuneration for regular and advanced skills in English that are mostly unused in the workplace, but which are taken as signals of other cognitive and non-cognitive skills by the employer. On the other hand, the same skill

²³ The model is estimated including interactions between English skills and the frequency of English usage at work, plus all the controls included in previous specifications (complete results are available upon request).

level is less remunerated when English is commonly used at work, because in this case having advanced competences in English may only represent a prerequisite to enter the job. In any case, the relevance of English skills as human capital remains clear given the positive returns to English knowledge — regardless of the frequency of its use at work — and general increases with added proficiency. Additionally, taking into account the precision of the estimated returns by frequency of English use at work, we are unable to discriminate against constant returns for each skill level in English.

5.3.2 Returns to English Skills by Age-Cohorts

Secondly, Table 8 shows the results when the sample is split into two subsamples by age cohort. These cohorts are a younger (25-39) cohort and an older (40-65) cohort. Consistent with the results from India reported by Azam et al. (2013), the return to English skills in Turkey appears to be significantly higher for the older cohort of workers. This evidence indicates that while the demand for workers with English language competence increased, the supply must have also increased since otherwise returns would be higher for the young than their elders. Indeed, in India, the older cohort of workers endowed of advanced English competences obtain up to 70% higher earnings than their counterparts from the same age-cohort who do not know English at all. However, advanced English skills are less well remunerated in the labor market for the younger cohort of workers, although the estimated premium is still positive and significant. The return to regular English skills is also higher for the older cohort, but not statistically different from that of the younger cohort. Overall, the evidence obtained by splitting the sample by cohort highlights the scarcity of this alternative human capital asset (i.e. English language competences) among the older subsample of workers. Additionally, controlling for occupation fixed-effects generates a more sensible reduction of the return to advanced English skills among the younger cohort. This result suggests that the effect of English knowledge on the chances of attaining a better remunerated job is especially pronounced in the earlier phases of the labor market career. An alternative explanation²⁴ for this evidence is that FL competences are becoming more relevant for acceding certain types of jobs

²⁴ This ambivalent interpretation of the results by cohort concerning occupation derives from the impossibility of separating age from (pure) cohort effects in a cross-section of data.

which are better rewarded than in the past. Finally, while basic English skills are not rewarded among the younger cohorts, there is a modest return for basic competences for the older cohort, which appears to come mainly from the occupational channel (i.e. it loses significance once we control for occupation fixed effects).

5.3.3 Returns to English Skills by Education and Occupation

Third, we consider the existence of potential complementarities between English competences and other labor market skills. Following, among others, Lang and Siniver (2009), Casale and Posel (2011) and Azam et al. (2013), we estimate separate equations for workers with low, medium and high educational attainments. The main results are reported in Table 9. The evidence for Turkey appears — to some extent — at odds with what is generally reported for other countries. While other authors obtained significant complementarities between education and FL knowledge (i.e. the returns to FL knowledge are higher for the more educated), in our case there is instead some weak evidence of substitutability between English skills and formal education. More specifically, the return to advanced English skills is similarly higher for the medium-and-low-educated workers, and the return to regular English competences is higher for low-educated individuals than for their medium-and-high-educated counterparts — who get a similar return to this level of command of English. Additionally, only low-educated workers obtain a positive reward for basic English skills. However, due to loss of precision, especially among the low-educated, returns to English skills are not statistically different between the three educational groups. Controlling for occupation reduces a substantial amount of the return to advanced English skills among workers with secondary education, although it barely modifies the return to the same level of competences among the low-educated. Finally, the positive return to basic English skills among low-educated workers is clearly reduced once occupation is controlled for and it further loses statistical significance after the inclusion of parental education.

In order to have a deeper insight about the (in)existence of language- formal education complementarities, we compute the return to English skills according to birth-cohort and

completed education (in a similar fashion as in Azam et al., 2013). The results, reported in Table 10, indicate complementarity between English skills and formal education among the older cohort, since the return to advanced English skills appears to be higher for medium and high educated workers — with a higher coefficient for the former than for the latter. However, among younger workers there is substitutability between formal education and English proficiency, since the return to advanced English skills is higher for the low-educated workers—given the (marginally) significant negative interaction coefficients between English competences and education dummies. On the one hand, this picture is consistent with the idea that, among younger workers, English knowledge is more widespread and also more commonly demanded while hiring the highly educated and hence not reflected in higher earnings. Still, less educated young workers fluent in English enjoy a comparative advantage relative to their equally-less-educated counterparts who do not have any English competence. On the other hand, among the older workers, only those who attained a certain level of formal education are able to exploit all the labor market potential of English proficiency.

Additional evidence on the complementarities between FL knowledge and other labor market skills can be obtained by estimating the model for different types of occupations, as done by Saiz and Zoido (2005) for the USA, by Lang and Siniver (2009) for Israel, and by Willams (2011) for EU countries. We divided the sample according to the standard high/low skill—white/blue collar categorization based on the two-digit ISCO88 occupational classification. In this case we obtain a rather limited degree of heterogeneity in the estimated return to English skills, which appears to be very similar for the four occupational groups (see Table 11). The most noticeable exception is the high return to the English skills for the high-skilled blue-collar workers compared to other workers, especially for advanced competences, although the estimates are somewhat imprecise and not distinguishable in statistical terms. It seems worth noting that the premium for being proficient in English is also positive, albeit imprecisely estimated, for low-skilled blue-collar workers, consistent with evidence regarding heterogeneous returns by education.

²⁵ Specifically, with the aim of maintaining a sufficient number of observation in each model, we estimated the equation(s) separately for the younger (25-39) and the older (40-65) birth-cohorts and we interacted English skills dummies with two dummies for completed education that capture the differential returns to English skills for medium-and-high educated workers (relative to the base category of low-educated workers).

5.3.4 Returns English skills by Rural/Urban Areas

The final point consists in estimating the model separately for urban and rural areas, as was done by Azam et al. (2013). They found that, among Indian male workers, there is no difference in returns to English language based on either rural or urban residential location. Similarly, our results for Turkish men in Table 12 show that returns to regular English skills is virtually the same for urban and rural areas, albeit that the earnings premium to advanced English skills is slightly higher in urban areas. This evidence is probably due to that, while demand for English skills should be higher in urban areas, most of the economic activities in which English is relevant and remunerated as an asset also take place in urban areas as this is where multinational firms, government and information and communication technology (ICT) intensive firms mostly operate. The presence of more schools in big cities and the increasing migration of more skilled workers towards urban agglomerations means that the supply of workers with English skills would be higher in such locations. In any case, the coefficients for workers in rural areas are somewhat imprecisely estimated. This does not provide evidence against the null hypothesis of equal returns to English skills between urban and rural areas.

Overall, our heterogeneity exercise indicates some heterogeneity in the returns to English skills, which in some cases appear to be opposite of what was previously reported in the literature. However, most of the differences observed in the point estimates are not statistically significant, stressing the robustness of positive economic returns to English competences in the Turkish labor market.

5.4 An IV strategy for the return to English skills

The results reported in the previous sections show that, generally, FL knowledge is associated with higher labor market earnings. More specific results regarding the return to English skills are extremely robust, indicating that the conditional relationship between competences in English and earnings is positive and substantial, confirming the evidence

obtained in other studies for both developed (Saiz and Zoido, 2005, Lang and Siniver, 2009, Ginsburg and Prieto-Ridríguez, 2011) and developing countries (Toomet 2011, Casale & Posel 2011, Azam et al. 2013). However, there might be some concern about the extent to which this conditional association is (not) close to the true causal parameters of interest. Previous evidence suggests that the return to FL knowledge appears to be barely — if at all — affected by endogeneity bias. In other words, the return to FL skills is still positive when estimated using either individual fixed-effects or instrumental variables methods. The stability of returns to English skills given the simultaneous inclusion of indicators for competences in other languages and also to several heterogeneity exercises make us feel comfortable that the extent of bias coming from unobserved characteristics should be rather limited. Even so, the presence of endogeneity bias might still present an issue meriting additional consideration. Moreover, our estimates might also be biased due to potential measurement/misclassification error in the self-reported measures of language skills, as there is a tendency to over-report²⁶. Further, there may be reverse causality since individuals who earn more can afford higher expenses for learning FLs, for the purposes of either work or leisure activities.

On the basis of these concerns, we implement an Instrumental Variable (IV) method with the aim of obtaining consistent estimates of the economic value of English competences in the Turkish labor market. In order to identify the key parameters (i.e. the return to English skills), at least one valid exclusion restriction is needed. The challenge of finding a suitable exclusion restriction is often very hard in the absence of quasi-experimental data, especially because of the non-trivial condition of orthogonality between the instrument and earnings potential. In this application, we exploit information on the frequency of English use for leisure as the exclusion restriction. We assume that the increase in the frequency of language use for leisure purposes 1) increases the propensity of being more skilled in English and 2) is related to labor market earnings only through its direct effect on English competences (i.e. it is conditionally unrelated to the error term of the earnings equation). The validity of condition 1) can be directly inferred from the data.

²⁶ See Dustmann and van Soest (2001, 2002, 2004) and Dustmann and Fabbri (2003) for detailed discussion about measurement error issue in the context strictly related to the earnings return to host country's language proficiency among immigrants.

The second hypothesis cannot be directly tested. Moreover, it might be argued that using English more frequently in the daily life would facilitate the access to better social networks and increase the chances of obtaining a better-remunerated job in which English represents a valuable asset. In this case, the second assumption would not be valid. However, we consider that controlling for the frequency of English use at work in the earnings equation would break this potential link between our exclusion restriction and unobserved earnings potential²⁷. Therefore, in what follows we present different specifications of our IV estimation that includes as additional control, the frequency of English usage at work. Moreover, we also provide the results from overidentification tests, which indicate whether the selected instrument can be reasonably excluded from the earnings equation(s).

Selected estimates from our IV strategy are reported in Table 13 (complete results are available upon request). Before discussing the results, it is worth remarking that our dependent variable (i.e. net monthly earnings) is only observed in intervals. Given that standard IV methods applied to the mid-point of earnings intervals might be seriously biased, we implement the "Instrumental Variable Interval Regression" (originally proposed by Bettin and Lucchetti, 2012), which can be estimated by Limited Information Maximum Likelihood (LIML). In order to simplify the model, we initially focus on an endogenous dummy of English proficiency, which takes the value of one for individuals who have advanced skills in English. The results are reported in the first four columns of Table 13.

In the bottom panel, we show the coefficients of the exclusion restrictions used in the first stage²⁸, whose estimates are in the expected direction. Moreover, the tests for

²⁷ Put in other words, if speaking English more frequently with friends, relatives and in the daily life in general provides access to jobs in which English is more important (and in principle used more often), this would be mostly picked up by the included dummies for the frequency of English use at work. Notice that just including dummies for the frequency of English usage at work in a standard earnings equation augmented by English skills produces a modest reduction in the estimated return to English knowledge, which is virtually the same as what we obtained controlling (only) for occupation fixed effects. Moreover, we applied our IV strategy also using as exclusion restriction the general frequency of FL use rather than English use, obtaining similar results (available upon request).

²⁸ The full results from the first stage regressions are not reported for space reasons, but are available upon request. The estimates indicate — in a robust way across different specifications — that having more schooling makes more likely achieving a higher level of competences in English. Experience affects negatively English skills, capturing the detrimental effect of age on the likelihood of mastering English. As expected, the increase of the frequency of English use at work and for leisure has a positive effect on English skills, with a more marked effect of the latter, while those who reside in urban

instrument validity suggest that the selected exclusion restrictions are strong predictors of language proficiency (i.e. they are jointly significant at any significance level). The overidentification test in the first column, in which we do not control for the frequency of English use at work, does not reject the null hypothesis of excludability only at a significance level of 10%. However, when the frequency of English use at work is additionally controlled for (second column), the frequency of English use for leisure appears to be conditionally independent from earnings and thusly can be correctly used as an exclusion restriction. Moreover, returns estimated using this IV strategy, and also controlling for English use at work, are again positive and statistically significant, ranging between 0.44 and 0.5. Indeed, these show little statistical difference from the estimate obtained without accounting for the endogeneity of English skills. Indeed, the exogeneity test indicates that the estimated returns to English skills obtained without controlling for potential endogeneity seem to be consistent, at least when the frequency of language use for leisure is included into the model with the linear first-stage.

In the subsequent step, we explicitly take into account the dichotomous nature of the endogenous variable (i.e. language proficiency). The LIML method proposed by Bettin and Lucchetti (2012) is in principle consistent also when the endogenous variable is neither a dummy, nor a continuous variable. However, we accommodate for a Probit specification for the first stage equation²⁹. The results from this alternative specification are reported in columns 5-8 of Table 13, and appear to be very similar to those obtained using a linear probability model for the first step. The most noticeable differences are the modest increase in the estimates' precision achieved using the Probit model for the first stage (also estimated by LIML), and the more pronounced decrease in the return to English proficiency when controlling for occupation and for parental education. Finally, we also implement a more compelling endogenous specification for the return to English skills, considering four increasing levels of English skills — i.e. we specify an Ordered Probit for the first step. The coefficients estimated allowing for endogenous (categorical) skills in English (columns 9-

areas have higher propensity to be proficient in English. Parental education also exerts a positive effect on the likelihood of having higher English skills. Finally, there is no significant effect of labor market variables and occupation FE.

The estimations have been carried out using the STATA routine "cmp" developed by Roodman (2011). For the IV estimation with a Linear Probability Model in the first stage we obtained the same results using "cmp" as with using the GRETL routine that has been kindly provided by Bettin and Lucchetti (2012) — which we also used for obtaining the overidentification tests.

12 of Table 14) are somewhat higher than those obtained without controlling for endogeneity (see column 1 of Table 6 for comparison), but confirm the existence of increasing returns to higher levels of competences in English. Also in this case, controlling for occupation and parental education reduce to a certain extent the point estimates, but does not undermine the significance and the general results obtained for regular and advanced English skills.

In general, our IV strategy confirms the reliability and the robustness of a positive and increasing economic value of English skills in the Turkish labor market. There is some modest sign of negative endogenous selection, given that the correlation between the error term of the earnings equation and error term of the first stage equation (i.e. the "rho" coefficient) is always negative and appears to be significant in non-linear first stage IV models. This means that, in principle, the return to English skills estimated neglecting endogeneity would be — at least to some extent — downward biased. However, the exogeneity test tends to lose statistical significance when controlling for occupation suggesting that, if any kind of endogenous selection into English skill levels exists, it would mainly operate through the occupational channel. Moreover, the "net" downward bias of the return to language skills has been previously attributed to the prevalence of the attenuation bias coming from measurement error in self-reported language proficiency over the unobserved ability bias (as suggested by Dustman and van Soest, 2001, 2002, and by Ginsburg and Prieto-Rodríguez, 2011 among others). Nevertheless, under non-classical errors-in-variables, which could be the case of our categorical language skills variable, the IV estimation should also provide upward biased coefficients, which represent upper bounds of the unbiased estimates (as shown by Kane et al., 1999 and Black et al., 2000). Apart from that, if our hypothesis about the exclusion restriction used in the IV estimation fails, the instrument (i.e. the frequency of English use for leisure) is correlated with the same unobservable elements affecting English competences and earnings potential and the return to English skills estimated with our IV models is likely to be biased towards the OLS (interval regression in our case) estimate. This would be especially likely in the case that unobserved heterogeneity mostly affects the propensity to know English and not too much the levels of proficiency, since the variation generated by the frequency of language use for

leisure affect only individuals who actually know some English. However, although it might be argued that the returns to English skills reported in this study may not represent the true causal parameters of interest, they might be still upwardly biased—at least to some extent, we cannot discard the existence of a positive and substantial economic value of English knowledge. Whatever the case may be, the overall results from our analysis can be considered as sufficiently robust as to conclude that skills in FLs and especially in English are positively rewarded in the labor market in the case of the Turkish economy.

5.5 Summary of Empirical Results

The aim of this paper is to quantify the returns to competences in FLs in Turkey. We initially consider the economic value of different FLs among employed males aged 25 to 65. Our results highlight that, in general, the knowledge of FLs has a positive economic value in the Turkish labor market. These returns appear to be (only) in part related to the occupational channel (i.e. those who master FLs are likely to be attracted into better paid occupations³⁰). The results are generally robust to the inclusion of controls for parental education, which proxy for both cognitive-and-non-cognitive skills and social networks. Among the more common languages in Turkey, English competences clearly represent a valuable asset, whose earnings return is robust across several specifications. The knowledge of Russian, especially advanced knowledge, is also highly rewarded in the labor market, as this language is relatively uncommon in Turkey. There is also some evidence of positive labor market rewards for mastering either French or German, although the economic value of these two languages seems mostly linked to occupation rather than productivity within occupations. On the contrary, knowing either Bulgarian or Arabic seems not to be rewarded in the labor market.

-

³⁰ Modeling the complex relationship between English knowledge, occupation and earnings represents an interesting extension of the current work, since the existing evidence concerning language proficiency among immigrants highlights that (immigrant) workers self-select into occupations according to their language skills, and this mediates a substantial part of the relationship between language and earnings (see Aldashev et al., 2009 and Chiswick and Miller, 2010). Moreover, the presence of salary premium for public sector workers according to their competences in English might introduce some positive language-related self-selection of more productive workers into the Turkish public sector, which can be treated in a similar fashion to Di Paolo (2012) for the case of Catalan knowledge and public/private sector selection.

In general, the earnings premiums for FL knowledge are comparable, but somewhat lower, than the returns to different levels of education in Turkey provided in Tansel (1994, 1996, 2010). Moreover, since English appears to be the most common FL spoken in Turkey, as well as in other non-English speaking countries, and has become the *lingua* franca for commerce and trade in the globalized world, we performed several additional estimations aimed at checking for heterogeneous returns to different levels of skills in this language. The earnings return to English skills obtained for Turkey is completely consistent with those for other developed and developing countries. In fact, the European evidence³¹ obtained by Ginsburgh and Prieto-Rodríguez (2011) suggest that returns to English knowledge varies from 10% in Denmark, where English is widely spoken, to 49% in Spain, where speaking English as FL is significantly less common. Regarding developing countries, Toomet (2011) reports a return to English skills of about 45% in Estonia and 62% in Lavtia. The two existing studies for the case of South Africa indicate an earnings return to English proficiency that range between 18-25% (Levinshon, 2004) and 41-44% (Casale and Posel, 2011). Finally, the results reported by Azam et al. (2013) show a 35% premium for advanced English skills (in their most complete specification) for Indian males.

As expected, and in line with the literature, we find that the return to English proficiency is higher for the older cohort and in urban areas. However, our results regarding language-skills versus education complementarity are somewhat at odds with the evidence obtained from other countries. In fact, Lang and Siniver (2009) indicate that the return to English knowledge in Israel is about 16% for high educated workers and only 5% for the less-educated group. Similarly, the results obtained by Casale and Posel (2011) show that the premium to English proficiency in South Africa is substantially higher for tertiary educated workers than for less educated individuals. Finally, Azam et al. (2013) obtain certain evidence in favor of complementarity between English skills and formal schooling, which is mostly driven by the results for more educated young workers. On the contrary,

_

³¹ Also Willams (2011) obtained significant and positive returns to the use of English at work in several European Countries. He also highlighted that the use of other languages — especially French and German — is relevant in some country, which can be considered in part consistent with our results regarding these two languages. However, we do not rely on the strict comparability between our results and those reported by Willams (2011), since he considered the return to language use at work instead of language competences.

our results suggest that the premium to advanced English skills is higher for medium-and-low educated workers, and returns to regular English skills are also higher for low educated workers, who also obtain a return to basic English competences. Moreover, on the one hand, we also find that the returns are higher at the low levels of education for the younger cohort of workers, suggesting some weak substitutability between formal education and English proficiency among young individuals. On the other hand, the return to advanced English skills increases with attained education for older workers, indicating complementarity between schooling and English among those individuals. In addition, we also explored the possibility of heterogeneous returns to English skills by occupational groups, although these results were less conclusive.

In any case, our heterogeneity analysis reveals that a positive economic value of English skills exists for several subgroups of workers, highlighting the overall significance of our results. With the aim of verifying that our estimates of the return to English skills are not just reflecting unobserved individual heterogeneity, we also implemented an IV strategy, based on information about the use of English skills for leisure and at work. The results obtained using different specifications — in which we account for the intervalcoding of earnings and for the discrete/ordinal nature of English skills — are very similar to those obtained without considering the potential bias provoked by unobserved heterogeneity and/or misclassification errors in self-reported English competences. This confirms the existence of increasing economic returns to different levels of English skills, which appears to be robust to the estimation method. Although it might be argued that the estimated returns to English knowledge and competences, as well as to other FLs, reported in this study do not exactly represent the true causal parameters of interest, the whole evidence reported in this paper suggest that we cannot discard a positive and substantial reward of this alternative form of human capital in the Turkish labor market.

7. Conclusions and Policy implications

The knowledge of foreign languages represent a form of human capital. Drawing on data from the 2007 Adult Education Survey, this is the first study that estimates the earnings returns to FL skills in Turkey, a country recently characterized by rapid economic and social development. The ongoing changes in the Turkish economy have fostered the relevance of and demand for FL competences in the labor market. However, the endowment of FL skills among the Turkish labor force appears to be rather scarce. Overall, this situation points to the existence of substantial economic premiums to the command of FLs. Quantifying such returns represents the main purpose of this paper.

Examining the returns to FLs is important, since it will guide policy makers and individuals about how much to invest in fostering competences in FLs among current and future generations of workers. Overall, the results from our study suggest that acquiring competences in FLs represents a profitable investment in the Turkish labor market. The returns to this investment are clearly positive at the individual level. Indeed, becoming proficient in English, but also in Russian and, to a lesser extent French, and German, constitutes a significant potential for higher earnings and, more generally, for better labor market performance, as FL knowledge seems to increase the chances of obtaining a better and more remunerated job. Thus proficiency in FLs has important implications in terms of labor market outcomes, since it improves employability, occupational prospects and earnings potential. Moreover, it seems plausible that the economic value of FL knowledge would be positive not only at the individual level, but also at the societal level.

Several researchers commented on the low level of human capital of Turkish workers, especially those employed in the informal sector³². As the Turkish AES-2007 data highlights, almost 60-65% of the Turkish labor force has only 8 years of education at the

³² The differential role of language skills in formal-and-informal sectors is an issue that should be examined in more detail, especially in the light of the relevance of informal employment in the Turkish labor market (as reported by Tansel and Kan, 2012). Unfortunately, the AES data do not provide suitable information for identifying informal workers (such as Social Security coverage), which prevented the in-deep analysis for formal and informal workers. However, we run separate language-augmented earnings regressions by type of employment (i.e. salaried workers, employees and self-employed workers) obtaining similar results for the return to English skills. This result can be taken as indirect evidence that being proficient in English should be rewarded in both the formal and informal sectors; in any case, a more detailed investigation of this issue should be done once more detailed data becomes available.

current compulsory, basic education level. The performance of the Turkish 15 year-olds in the PISA³³ test is rather poor. Indeed, in the 2009 PISA test Turkey ranks 32nd among the 34 countries ahead of only Chile and Mexico. The average 15-year-old student in Turkey is one full year behind the OEDC average (World Bank, 2013). Enhancing human capital, the endowment of education, and its equitable distribution among different socio-economic groups present current challenges for achieving and maintaining a sustainable path of growth and development in the mid- to long-term in Turkey. However, our results suggest that fostering FL skills should be taken as an additional challenge for Turkish policy makers. There are several reasons to consider that increasing competences in FLs among the Turkish population would further promote international trade, internationalization and openness in the Turkish economy, as well as R&D activities and innovation. In turn, this would generate greater potential for growth and socio-economic development of the nation, improving its position in the global knowledge economy.

Indeed, improving English skills among the population would be especially beneficial for a mid-sized developing country such as Turkey, since it may help reduce existing disparities in global competition between emerging economies for international trade and attracting new FDIs. This is extremely relevant in light of the significant scale and resource advantages of the two leading Asian emerging countries, India and China. In fact, in the former, English represents a former colonial language that is co-official and widely spoken among the population, especially among the highly educated, and the latter has the largest English-learner population in the world (Crystal, 2008, He and Li, 2009). Moreover, we believe that, relative to other mid-sized emerging economies, fostering competences in English, as well as in other relevant European languages, might be especially important in Turkey for two additional reasons. First, given the geographical location of the country, this could favor its historical role of "bridge" for commodities trading between Asia and Europe. Second, reducing language barriers would be especially relevant for further attenuating already reduced cultural barriers between Turkey and EU countries, which might represent an additional stimulus for commerce and trade.

⁻

³³ PISA stands for Programme for International Student Assessment. It is implemented each three years (from 2001) by OECD to test 15- year-olds skill and knowledge and competencies in the areas of reading, mathematics and science.

Several policy implications can be directly advocated in light of our results, which can be reasonably extrapolated to other developing countries as well as to developed countries with insufficient endowment of FL skills in their labor forces. First, policy makers should emphasize teaching of English at schools, in order to increase the English proficiency of future generations of workers. This would be especially important due to growing demand for FLs competences in the Turkish labor market in the near future, with the prospect of further economic growth and development and possible access to the EU. The 1997 Turkish Education System reform increased the amount of FL teaching during the schooling process. The radical changes of the subsequent 2012 reform also introduced a gradual increase in FL instruction. However, there is no evidence about the effectiveness of these reforms in improving the FL proficiency of students from different grades. Therefore, evaluating the effect of the 1997 reform on English proficiency represents an object of our future research³⁴. Moreover, the government should also foster English teachers' training and professional requirements, since teachers play a fundamental role in guaranteeing the effectiveness of the above-mentioned educational reforms.

Second, for the current generation of workers, future public policies should be directed to encourage and subsidize their attendance at private FL centers. This is a sensible approach as our findings point out certain, albeit not high, substitutability between English skills and general schooling for the young. In fact, beyond earnings, FL skills may also enhance employability and labor market opportunities for low-educated young individuals who may possibly come from a disadvantaged socio-economic and family background. Moreover, as suggested by Rupérez-Micola et al. (2012), broadcasting films or programs in their original English with subtitles in national language, as done in several countries, especially in Northern Europe, might help increase English skills among the population.

Here we emphasize English as the FL to be taught, not only because of its international value and in light of its relevance compared to other languages that emerges from our results, but also because there currently is a substantial stock of English language teachers,

_

³⁴ Indeed, the future availability of the Turkish AES 2012 data will enable estimating the causal effect of the increase in teaching English at schools with the 1997 reform. In fact, the new data will contain information about individuals who are affected by the reform (i.e. the treatment group) and the others who are not exposed to the reform (i.e. the control group).

albeit still less than demand for them. Teaching Russian in schools would take time to accomplish, because of the need first to train teachers. The current demand for Russian speaking workers could be met by teaching Russian at special schools such as tourism schools or at FL centers. There may be also some space for policies aimed at improving competences in German and French. However, our less conclusive results regarding these two languages, and given the hegemony of English as the *lingua franca*, less priority should be given to investments in these languages.

In contrast, our findings suggest that there is no earnings premium to knowledge of Arabic and Bulgarian in the Turkish labor market. Given these results, from an economic perspective, the policy makers should discontinue investing scarce resources into teaching Arabic at the religious vocation schools³⁵. These skills are not rewarded in the labor market and hence are non-productive. Also noteworthy is the absence of Chinese language instruction in Turkey, excepting a couple of university programs. Chinese language instruction could be important given recent increases in the volume of trade with China³⁶.

Further, since most productive potential of FL skills is expected to be allocated in the private sector, especially among firms exposed to English-intensive activities such as international trade, R&D, ICT and tourism, private businesses should contribute to financing FL training in their workforce and complement government's public investment. Finally, Turkey should be able to benefit more from language competences of citizens with immigrant backgrounds, such as the growing population of return-migrants from Germany attracted by the current economic development path of the country.

2

³⁵ A recent law passed at the parliament mandated that starting with the 2013-2014 academic year the "Ottoman language" will be a compulsory course at the social sciences high schools and an elective course in all other high schools (Sol Portal, April 6, 2013). A dead language like Ottoman language is expected to have no economic value in the labor market. It could be instructed to those specializing in the Ottoman history or Ottoman literature at the undergraduate or post-graduate programs of the universities rather than at the high schools.

³⁶ China was Turkey's 14th largest export partner in 2012 and 1st export destination among Asia-Pacific countries. Turkey's exports to China were 2.8 billion USD in 2012 and 2.5 billion USD in 2011. Regarding imports, China was Turkey's 3rd largest partner in 2012 and Turkey's imports from China realized as 21.3 billion USD in 2012, and 21.7 billion USD in 2011. Moreover, FDI stock of China in Turkey amounts to 26 million USD between 2002 and 2012 (TURKSTAT, 2013). Therefore, we expect a substantial and growing labor market value of Chinese language competences in Turkey. Unfortunately, we were unable to quantify the return to Chinese knowledge, since the number of Chinese speakers in our sample was too low in order to consider this language in the empirical analysis.

References

Adak, M., 2010. Foreign trade and economic growth: The case of Turkey. Middle Eastern Finance and Economics. 8, 137-145.

Adesope, O. O., Lavin, T., Thompson, T., Ungerleider, C., 2010. A systematic review and meta-analysis of the cognitive correlates of bilingualism. Review of Educational Research. 80(2), 207-245.

Adsera, A. and Pytlikova, M., 2012. The role of language in shaping international migration. Institute for the Study of labor Working Paper Series No: 6333, Bonn.

Albouy, D., 2008. The wage gap between Francophones and Anglophones: A Canadian perspective, 1970–2000. Canadian Journal of Economics / Revue Canadienne d'Economique, 41(4), 1211-1238.

Aldashev, A., Gernandt, J., Thomsen, S. L., 2009. Language usage, participation, employment and earnings: Evidence for foreigners in West Germany with multiple sources of selection. Labour Economics. 16, 330-341.

Angrist, J. D., Chin, A., Godoy, R., 2008. Is Spanish-only schooling responsible for the Puerto Rican language gap? Journal of Development Economics. 85(1-2), 105-128.

Angrist, J. D., Lavy, V., 1997. The effect of a change in language of instruction on the returns to schooling in Morocco. Journal of Labor Economics. 15(1), S48-76.

Angrist, J. D., Pischke, J. S., 2008. **Mostly Harmless Econometrics: An Empiricist's Companion,** Princeton University Press, New Jersey.

Arslanturk, Y., Balcilar, M., Ozdemir, Z. A., 2011. Time-varying linkages between tourism receipts and economic growth in a small open economy. Economic Modelling, 28(1-2), 664-671.

Aydın, Y., 2012. Emigration of highly qualified Turks: A critical review of the societal discourses and social scientific research, in: Paçacı-Elitok, S., Straubhaar, T. (Eds), Turkey, Migration and the EU: Potentials, Challenges and Opportunities, Edition HWWI, Institute of International Economics (HWWI), 5(5), Hamburg.

Azam, M., Chin, A., Prakash, N., 2013. The returns to english-language skills in India. Economic Development and Cultural Change. 61(2), 335-367.

Benito, G. R., Gripsrud, G., 1992. The expansion of foreign direct investments: Discrete rational location choices or a cultural learning process? Journal of International Business Studies. 23(3), 461-476.

Berman, E., Lang, K., Siniver, E., 2003. Language skill complementarity: Returns to immigrant language acquisition. Labour Economics. 10(3), 265-90.

Bettin, G., Lucchetti, R., 2012. Interval regression models with endogenous explanatory variables. Empirical Economics. 43(2), 475-49.

Bialystok, E., 1999. Cognitive complexity and attention control in bilingual mind. Child Development. 70(3), 636-644.

Black, D. A., Berger, M. C., Scott, F. A., 2000. Bounding parameter estimates with non-classical measurement error. Journal of the American Statistical Association. 95(451), 739-748.

Bleakley, H., Chin, A., 2004. Language skills and earnings: Evidence from childhood immigrants. The Review of Economics and Statistics. 86(2), 481-96.

Bleakley, H., Chin, A., 2010. Age at arrival, English proficiency, and social assimilation among US immigrants. American Economic Journal: Applied Economics. 2(1), 165-92.

Casale, D., Posel, D., 2011. English language proficiency and earnings in a developing country: The case of South Africa. The Journal of Socio-Economics. 40(4), 385-393.

Cattaneo, A., Winkelmann, R., 2005. Earning differentials between German and French speakers in Switzerland. Swiss Journal of Economics and Statistics. 141(2), 191-212.

Cebeci, T., Fernandes, A. M., 2013. Micro dynamics of Turkey's export boom in the 2000s. The World Bank. Policy Research Working Paper No: 6452, Washington, DC.

Chiswick, B. R., 2008. The economics of language: An introduction and overview. Institute for the Study of Labor (IZA) Discussion Paper No: 3568, Bonn.

Chiswick, B. R. Miller, P. W., 1995. The endogeneity between language and earnings: International analyses. Journal of Labor Economics. 13(2), 246-88.

Chiswick, B. R. Miller, P. W., 2007. **The Economics of the Language: International Analyses**, Routledge, New York.

Chiswick, B. R., Miller, P. W., 2009. Earnings and occupational attainment among immigrants. Industrial Relations. 48(3), 454-465.

Chiswick, B. R., Miller, P. W., 2010. Occupational language requirements and the value of English in the US labor market. Journal of Population Economics. 23, 353-372.

Chiswick, B. R., Miller, P. W., 2013. The impact of surplus skills on earnings: Extending the over-education model to language proficiency. Economics of Education Review. 36, 263-275.

Church, J., King, I., 1993. Bilingualism and network externalities. Canadian Journal of Economics. 26, 337-245

Cooper, T. C., 1987. Foreign language study and SAT-Verbal Scores, The Modern Language Journal. 71(4), 381-387.

Crystal, D., 2008. Two thousand million? English Today. 24(1), 3-6.

Çetinkaya, M., Erdoğan, S., 2010. VAR analysis of the relation between GDP, import, and export: Turkey case. International Research Journal of Finance and Economics. 55, 135-145.

Di Paolo, A., 2011. Knowledge of Catalan, public/private sector choice and earnings: Evidence from a double sample selection model. Hacienda Pública Española/Review of Public Economics. 197(2), 9-35.

Di Paolo, A., Raymond, J. L., 2012. Language knowledge and earnings in Catalonia. Journal of Applied Economics. 15(1), 89-118.

Dustmann, C., 1994. Speaking fluency, writing fluency and earnings of migrants. Journal of Population Economics. 7(1), 133-56.

Dustmann, C. Fabbri, F., 2003. Language proficiency and labour market performance of immigrants in the UK. Economic Journal. 113(489), 695-717.

Dustmann, C., van Soest, A., 2001. Language fluency and earnings: Estimation with misclassified language indicators. The Review of Economics and Statistics. 83(4), 663-674.

Dustmann, C., van Soest, A., 2002. Language and the earnings of immigrants. Industrial and Labor Relations Review. 55(3), 473-92.

Dustmann, C., van Soest, A., 2004. An analysis of speaking fluency of immigrants using ordered response models with classification errors. Journal of Business and Economic Statistics. 22, 312-321.

Education First, (2011 and 2012) Education First English Proficiency Index. (Available at www.ef.com/epi).

European Union, 2006. Europeans and their languages. Special Euro-Barometer 386/Wave EB77.1 (available at http://ec.europa.eu/languages/documents/2006-special-eurobarometer-survey-64.3europeans-and-languages-report en.pdf).

European Union, 2012. Europeans and their languages, Special Euro-Barometer 243/Wave 64.3 (available at http://ec.europa.eu/public opinion/archives/ebs/ebs/286 en.pdf).

Eurostat, 2012. Gross Domestic Expenditure on R&D as Percent Share of GDP (available at http://epp.eurostat.ec.europa.eu/statistics_explained/index.php/R_%26_D_expenditure).

Federal Statistical Office of Germany (2013) Accessed on October 9, 2013: (available at https://www.destatis.de).

Fidrmuc, J., 2011. The economics of multilingualism in the EU, in: Eger, T., Schäfer H-B. (Eds.), Research Handbook on the Economics of European Union Law. Edward Elgar, Cheltenham, pp. 331-352.

Fidrmuc, J., Fidrmuc, J., 2009. Foreign languages and trade. Centre for Economic Policy Research (CEPR) Discussion Paper No:7228, London.

Gabszewicz, J., Ginsburgh, V., Laussel, D., Weber, S., 2011a. Foreign languages acquisition: Self-Learning and language schools.

Gabszewicz, J., Ginsburgh, V., Laussel, D., Weber, S., 2011b. Bilingualism and communicative benefits, annals of economics and statistics. Annales d'Économie et de Statistique. 101/102, 271-286.

Galasi, P., 2003. Estimating wage equations for Hungarian higher education graduates. Budapest Working Papers on the Labour Market.

Ginsburgh, V. A., Laussel, D., Weber, S., 2011. Foreign language acquisition: Self-Learning and language schools. Review of Network Economics. 10(1), 1446-9022, DOI: 10.2202/1446-9022.1185.

Ginsburgh, V. A., Ortuno-Ortin, I., Weber, S., 2007. Learning foreign languages. Theoretical and empirical implications of the Selten and Pool model. Journal of Economic Behavior and Organization. 64, 337-347.

Ginsburgh, V. A., Prieto-Rodriguez, J., 2011. Returns to foreign languages of native workers in the EU, Industrial and Labor Relations Review. 64(3), 599-617.

Grin, F., Sfreddo, C., 1998. Language-based earnings differentials on the Swiss labour market: Is Italian a liability? International Journal of Manpower. 19(7), 520-532.

Grin, F., Sfreddo, C. Vaillancourt, F., 2010. **The Economics of the Multilingual Workplace**. Routledge, New York.

Gunduz, L., Abdulnasser-Hatemi, J., 2005. Is the tourism-led growth hypothesis valid for Turkey? Applied Economics Letters. 12(8), 499-504.

He, D, Li, D. C. S., 2009. Language attitudes and linguistic features in the 'China English' debate. World Englishes. 28(1), 70-89.

Henley, A., Jones, R. E., 2005. Earnings and linguistic proficiency in a bilingual economy. The Manchester School. 73(3), 300-320.

Hoon, C. O., Selmier, W. T., Lien, D., 2011. International trade, foreign direct investment, and transaction costs in languages. The Journal of Socio-Economics. 40(6), 732-735.

Howitt, P., 1999. Steady endogenous growth with population and R&D inputs growing. Journal of Political Economy. 107, 715-730.

Hutchinson, William K. 2005. Linguistic Distance" as a Determinant of Bilateral Trade. Southern Economic Journal. 72(1), 1-15.

Isphording, I., Otten, S., 2013. The costs of Babylon – Linguistic distance in applied economics. Review of International Economics. 21(2), 354-369.

Kane, T. J., Rouse, C. E., Staiger, D., 1999. Estimating returns to schooling when schooling is misreported. National Bureau of Economic Research (NBER) Working Paper No: 7235, Cambridge, MA.

Karabag, S. F., Tuncay-Celikel, A., Berggren, C., 2011. The limits of R&D internationalization and the importance of local initiatives: Turkey as a critical case. World Development. 39(8), 1347-1357.

Klein, C., 2003. La valorisation des compétences linguistiques sur le marché du travail luxembourgeois. CEPS/INSTEAD Working Paper No: 139, Luxemburg.

Kogut, B., Harbir, S., 1988. The effect of national culture on the choice of entry mode. Journal of International Business Studies. 19(3), 411-432.

Koru, S., Akesson, J., 2011. Türkiye'de ingilizce açığı, (English language deficiency in Turkey): Türkiye Ekonomi Politikalari Araştırma Vakfı Politika Notu No: N201157, Ankara.

Kotil, E., Konur, F., 2010. The relationship between growth and foreign trade in Turkey: A granger causality approach (1989-2007). Middle Eastern Finance and Economics. 1(6), 32-40.

Ku, H., Zussmann, A., 2010. Lingua Franca: The role of English in international trade. Journal of Economic Behavior and Organization. 75, 250-260.

Lang, K., Siniver, E., 2009. The return to English in a non-English speaking country: Russian immigrants and native Israelis in Israel. The B.E. Journal of Economic Analysis and Policy. 9(1), 1-30.

Leikin, M., 2012. The effect of bilingualism on creativity: Developmental and educational perspectives. International Journal of Bilingualism. 10(10), 1-17.

Leslie, D., Lindley, J., 2001. The impact of language ability on employment and earnings of Britain's ethnic communities. Economica. 68(272), 587-606.

Leslie, D., Russell, H., Forbes, A., 2002. Foreign language skills and tourism management courses in the UK. Industry and Higher Education. 16(6), 403-414.

Levinsohn, J., 2007. Globalization and the returns to speaking English in South Africa, in: Harrison, A. E., (Ed.), Globalization and Poverty. University of Chicago Press, Chicago, pp. 629-646.

Melitz, J., 2008. Language and foreign trade. European Economic Review. 52, 667-699.

Olsen, S., Brown, L. K., 1992. The relation between high school study of foreign language and ACT English and Mathematics performance. Association of Departments of Foreign Languages (ADFL) Bulletin. 23(3), 47-50.

Organization for Economic Co-Operation and Development (OECD) (2012) **OECD Economic Surveys: Turkey.** OECD Publications. Paris (DOI: 10.1787/eco surveys-tur-2012-en).

Özçelik, E., Taymaz, E., 2008. R&D support programs in developing countries — the Turkish case. Research Policy. 37(2), 258.275.

Öztürk, İ., Acaravcı, A., 2010. Testing the export-led growth hypothesis: Empirical evidence from Turkey. The Journal of Developing Areas. 44(1), 245-254.

Quella, N., Rendon, S., 2012. Occupational selection in multilingual labor markets: The case of Catalonia. International Journal of Manpower. 33(8), 918-937.

Rendon, S., 2007. The Catalan premium: Language and employment in Catalonia. Journal of Population Economics. 20(3), 669-686.

Roodman, D., 2011. Fitting fully observed recursive mixed-process models with cmp, The Stata Journal. 11(2), 159-206.

Rupérez-Micola, A., Bris, A., Banal-Estañol, A., 2012. TV or not TV? Subtitling and English skills. Department of Economics and Business, Working Paper No:1156, Universitat Pompeu Fabra (available at http://albertbanalestanol.com/wp-content/uploads/TV-or-not-TV-Subtitling-and-English-Skills.pdf).

Saiz, A., Zoido, E., 2005. Listening to what the world Says: Bilingualism and earnings in the United States. The Review of Economics and Statistics. 87(3), 523-538.

Seargeant, P., Erling, E., 2011. The discourse of English as a language for international development: Policy assumptions and practical challenges, in Coleman, H. (Ed.), Dreams and Realities: Developing Countries and the English Language. British Council, London, pp. 255-274.

Segerstrom, P., 2000. The long-run growth effects of R&D subsidies. Journal of Economic Growth. 5, 277-305.

Selten, R., Pool, J., 1991. The distribution of foreign language skills as a game equilibrium, in: Reinhard S. (Ed.), Game Equilibrium Models. Berlin: Springer-Verlag, Berlin, pp. 64-84.

Shapiro, D., Stelcner, M., 1997. Language earnings in Quebec: Trends Over Twenty Years,1970–1990. Canadian Public Policy. 23(2), 115-40.

Shields, M. A., Price, W. S., 2002. The English language fluency and occupational success of ethnic minority immigrant men living in English metropolitan areas. Journal of Population Economics. 15(1), 137-160.

State Planning Organization (SPO) (2006) Ninth Five-Year Development Plan 2007-2013. SPO Publications. Ankara.

Stewart, M., 1983. On least squares estimation when the dependent variable is grouped. Review of Economic Studies. 50(4), 737-5.

Tansel, A., 1994. Wage employment, earnings and returns to schooling for men and women in Turkey. Economics of Education Review. 13(4), 305-320.

Tansel, A., 1996. Self-employment, wage employment and returns to education for urban men and women in Turkey in education and the labor market in Turkey. in: Bulutay, T. State Institute of Statistics (SIS) Publication. Ankara, pp.175-208.

Tansel, A., 2010. Changing returns to education for men and women in a developing country: Turkey, 1994-2005, paper presented at the European Society for Population Economics (ESPE) Conference, London, UK and at the Middle East Economic Association (MEEA) Conference, March 2009 in Nice, France.

Tansel, A., 2012. 2050'ye Doğru nüfus bilim ve yönetim: İşgücü piyasasına bakış, TUSIAD and UNFPA. Publication No: Tüsiad-T/2012-11/536, İstanbul.

Tansel, A., Kan, E. O., 2012. The formal/informal employment earnings gap: evidence from Turkey. Institute for the Study of Labor (IZA) Discussion Paper No: 6556, Bonn.

Tansel, A. Ya şar, P. 2010. Macroeconomic impact of remittances on output growth: Evidence from Turkey. Migration Letters. 7(2), 132-143.

Toomet, O., 2011. Learn English, Not the local language! Ethnic Russians in the Baltic States. American Economic Review: Papers and Proceedings. 101(3), 526-531.

Tucci, I., Wagner, G. G., 2004. Foreign language skills- an important additional qualification in the services sector, Economic Bulletin. 41, 43-46.

Turkish Demographic Health Survey (TDHS), 2008. Turkish Demographic Health Survey, 2008. Hacettepe University Institute of Population Studies. Ankara.

Turkish Statistical Institute (TURKSTAT), (2012) Informal Education in Turkey, Turkish Statistical Institute Publication. Ankara.

Turkish Statistical Institute (TURKSTAT) (2010), Adult Education Survey, 2007. Turkish Statistical Institute Publication.No:3433. Ankara (available at http://www.tuik.gov.tr/Kitap.do?metod=KitapDetay&KT_ID=5&KITAP_ID=218).

Turkish Statistical Institute (TURKSTAT) 2013 (available at www.tuik.gov.tr).

United Nations Conference on Trade and Development (UNCTAD) 2013. World Investment Report. United Nations Publications. Geneva (available at http://unctad.org/en/PublicationsLibrary/wir2013_en.pdf).

Williams, D., 2011. The economic returns to multiple language usage in Western Europe. International Journal of Manpower. 32(4), 372-393.

World Bank, 2011. Improving the quality and equity of basic education in Turkey: Challenges and options. Human Development Department, Europe and Central Asia Region Report No: 54131, Washington DC.

World Bank, 2013. Promoting excellence in Turkey's schools. Human Development Sector Unit Europe and Central Asia Region Report No: 77722, Washington DC.

Yağmur, K., 2001. Languages in Turkey. in: Extra, G. and Gorter, D. (Eds.), The Other Languages of Europe. European Cultural Foundation, Clevedon, UK.

TABLES

Table 1: foreign languages among Turkish employed males

	no Foreign	one Foreign	two Foreign	three or	TOTAL
	Languages	Language	Languages	more FLs	(%)
BIRTH COHORT					
cohort 25-39	64.59	31.11	3.70	0.61	100
cohort 40-65	73.62	22.63	3.45	0.29	100
COMPLETED EDUCATION					
low-education	92.93	6.18	0.73	0.16	100
medium-education	57.83	37.93	3.83	0.40	100
high-education	18.35	68.41	11.69	1.55	100
OCCUPATION					
high-skilled white collars	45.39	46.66	7.06	0.89	100
low-skilled white collars	62.34	32.91	4.00	0.75	100
high-skilled blue collars	86.02	12.63	1.22	0.14	100
low-skilled blue collars	80.08	17.04	1.73	0.14	100
URBAN/RURAL AREAS					
urban	64.49	31.11	3.88	0.52	100
rural	78.66	18.12	2.89	0.34	100
TOTAL (%)	68.60	27.34	3.59	0.47	100
NUMBER OF OBSERVATIONS	6307	2514	330	43	9194

Table 2: spoken foreign language (among foreign language speakers, N = 2877)

	English	French	German	Arabic	Bulgarian	Russian	Others
BIRTH COHORT							
cohort 25-39	85.4	4	10.5	7.8	0.1	3.1	1.4
cohort 40-65	70.1	13	14	12.2	0.9	1.8	1.4
COMPLETED EDUCATION	N						
low-education	45.9	4.1	18	30	2.5	4.7	7.5
medium-education	79.9	7.2	10.8	8	0.2	3	1.1
high-education	88.6	8.3	11.4	5.8	0	1.6	0.1
OCCUPATION							
high-skilled white collars	83.8	8	11.8	7.7	0.2	2.3	0.7
low-skilled white collars	85.2	5.6	11.2	7	0	3.2	0.8
high-skilled blue collars	68.4	7	12.2	14.4	1	2.5	3.2
low-skilled blue collars	68.5	7.1	12.2	14.5	1	3.3	3
URBAN/RURAL AREAS							
urban	82.4	7.5	11.4	7.0	0.3	2.8	0.9
rural	69.2	6.7	13.5	19.5	0.9	1.8	3.2
TOTAL (%)	79.8	7.3	11.8	9.5	0.4	2.6	1.4

Table 3: foreign language skills (among foreign language speakers, N = 2877)

	% over FL speakers	basic skills	regular skills	advanced skills
first FL = English	76.86			
		55.02	31.82	13.16
first FL = French	5.23			
		73.51	16.56	9.93
first FL = German	8.31	40.15	22.00	10.55
C' TT A 1	6.00	49.17	32.08	18.75
first FL = Arabic	6.89	17.09	37.19	45.73
first FL = Bulgarian	0.24	17.09	37.19	43.73
ilist FL – Bulgarian	0.24	28.57	28.57	28.57
first FL = Russian	0.90	20.57	20.37	20.57
mst I L – Russian	0.70	11.54	69.23	19.23
first FL = Other	1.56		2, 120	
		20.00	22.22	57.78

Table 4: foreign languages and net monthly earnings (in Turkish liras)

	All the sample	at least one FL	English	French	German	Arabic	Bulgarian	Russian	Other FL
415 TL or less	19.0	7.3	5.6	5.6	7.3	19.8	27.3	2.7	27.5
416-529 TL	18.0	10.6	9.7	8.5	10.0	12.4	9.1	12.0	22.5
530-799 TL	22.0	17.9	17.9	11.3	19.6	15.7	36.4	13.3	25.0
800-1149 TL	21.1	27.5	27.6	28.3	25.2	34.1	9.1	33.3	7.5
more than 1150 TL	19.9	36.7	39.3	45.2	37.8	17.9	18.2	38.7	17.5
Total (%)	100	100	100	100	100	100	100	100	100
average monthly earnings	742.2	986	1018.7	1102.9	992.5	776.5	657.7	1030.2	615.4

Table 5: foreign language knowledge and earnings

Table 5: foreign language kno	Number									
	of foreign	English	French	German	Arabic	Bulgarian	Russian	all FLs	all FLs	all FLs
	languages									
constant	5.275 ^a	5.259 ^a	5.265 ^a	5.260 ^a	5.255 ^a	5.255 ^a	5.255 ^a	5.276 ^a	5.256 ^a	5.200 ^a
	(0.039)	(0.039)	(0.040)	(0.039)	(0.039)	(0.039)	(0.039)	(0.040)	(0.044)	(0.045)
years of schooling	0.073^{a}	0.074^{a}	0.080^{a}	0.080^{a}	0.081^{a}	0.081^{a}	0.081^{a}	0.073^{a}	0.052^{a}	0.049^{a}
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
potential experience	0.033^{a}	0.034^{a}	0.032^{a}	0.032^{a}	0.032^{a}	0.032^{a}	0.032^{a}	0.034^{a}	0.031^{a}	0.032^{a}
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)
(potential experience) ²	-0.001 ^a	-0.001^{a}	-0.000^{a}	-0.000^{a}	-0.000^{a}	-0.000^{a}	-0.000^{a}	-0.001^{a}	-0.000^{a}	-0.000^{a}
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
employee - permanent contract					reference c	ategory				
employee - fixed term contract	-0.246 ^a	-0.244 ^a	-0.243 ^a	-0.244 ^a	-0.243 ^a	-0.243 ^a	-0.244 ^a	-0.245 ^a	-0.225 ^a	-0.214 ^a
	(0.024)	(0.024)	(0.024)	(0.024)	(0.024)	(0.024)	(0.024)	(0.024)	(0.025)	(0.025)
self employed	-0.063^{a}	-0.064^{a}	-0.066^{a}	-0.066^{a}	-0.066^{a}	-0.066^{a}	-0.067^{a}	-0.064^{a}	-0.022	-0.019
	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.015)	(0.018)	(0.018)
employer	0.412^{a}	0.413^{a}	0.415^{a}	0.414^{a}	0.415^{a}	0.415^{a}	0.413^{a}	0.411^{a}	0.356^{a}	0.350^{a}
	(0.026)	(0.026)	(0.026)	(0.026)	(0.026)	(0.026)	(0.026)	(0.026)	(0.028)	(0.028)
urban area	0.253^{a}	0.250^{a}	0.252^{a}	0.252^{a}	0.252^{a}	0.252^{a}	0.251^{a}	0.249^{a}	0.154^{a}	0.150^{a}
	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)	(0.013)
parental education = no education	(0.010)	(0.012)	(0.012)	(0.010)	reference c	'	(0.012)	(0.010)	(0.010)	(0.010)
parental education no education					rejerence e					
parental education = primary										0.084^{a}
Farmer Farmer										(0.012)
parental education = secondary										$0.122^{\acute{a}}$
Facility and the second of the										(0.028)
parental education = tertiary										0.164^{a}
										(0.033)
occupation fixed effects	no	yes	yes							
number of observations	9194	9194	9194	9194	9194	9194	9194	9194	9194	9194
log-likelihood	-12929	-12931	-12956	-12955	-12958	-12958	-12953	-12923	-12524	-12494
pseudo R ²	0.206	0.206	0.204	0.205	0.204	0.204	0.205	0.207	0.237	0.237

Note: robust standard errors within parenthesis in italic. c Significant at p < 0.1, b significant at p < 0.05, a significant at p < 0.01.

Table 5 (continued): foreign language knowledge and earnings

	Number of foreign languages	English	French	German	Arabic	Bulgarian	Russian	all FLs	all FLs	all FLs
no foreign languages					reference	category				
one foreign language	0.091 ^a (0.015)									
two foreign languages	0.140^{a} (0.031)									
three or more foreign languages	0.317 ^a (0.076)									
knows English		0.112 ^a (0.015)						0.113 ^a (0.015)	0.090^{a} (0.015)	0.082^{a} (0.015)
knows French		(/	0.064 (0.039)					$0.080^{\rm b}$ (0.038)	0.066^{c} (0.037)	0.066^{c} (0.037)
knows German			(0.039)	0.064 ^b (0.028)				0.065^{b} (0.028)	0.041	0.036
knows Arabic				()	-0.019 (0.033)			-0.021 (0.033)	-0.057° (0.032)	-0.044 (0.032)
knows Bulgarian					(0.033)	-0.027 (0.168)		-0.035 (0.155)	-0.049 (0.178)	-0.056 (0.172)
knows Russian						(0.108)	0.186 ^a (0.056)	0.142 ^b (0.056)	0.146^{a} (0.053)	0.135 ^b (0.054)
occupation fixed effects	no	по	no	no	no	no	no	no	yes	yes
number of observations	9194	9194	9194	9194	9194	9194	9194	9194	9194	9194
log-Likelihood pseudo R ²	-12929 0.206	-12931 0.206	-12956 0.204	-12955 0.205	-12958 0.204	-12958 0.204	-12953 0.205	-12923 0.207	-12524 0.237	-12494 0.237

Note: robust standard errors within parenthesis in italic. c Significant at p < 0.1, b significant at p < 0.05, a significant at p < 0.01.

Table 6: return to foreign language skills

	English skills	French skills	German skills	Russian skills	all FLs skills	all FLs skills	all FLs skills
no foreign languages		511115		erence cate		52225	511115
basic English skills	0.030^{c}				0.041 ^b	0.034 ^b	0.030^{c}
_	(0.017)				(0.017)	(0.016)	(0.016)
regular English skills	0.198^{a}				0.211^{a}	0.173^{a}	0.165^{a}
	(0.024)				(0.024)	(0.023)	(0.023)
advanced English skills	0.454^{a}				0.468^{a}	0.390^{a}	0.378^{a}
	(0.038)				(0.038)	(0.037)	(0.038)
basic French skills		-0.030			0.039	0.034	0.036
		(0.051)			(0.051)	(0.049)	(0.049)
regular French skills		0.102			0.177^{c}	0.119	0.130
		(0.101)			(0.100)	(0.092)	(0.088)
advanced French skills		-0.024			0.055	-0.066	-0.092
		(0.151)			(0.150)	(0.135)	(0.137)
basic German skills			0.058		0.115^{b}	0.102^{b}	0.099^{b}
			(0.047)		(0.047)	(0.045)	(0.045)
regular German skills			-0.071		-0.015	-0.070	-0.073
			(0.055)		(0.054)	(0.053)	(0.053)
advanced German skills			0.105		0.157^{c}	0.143^{c}	0.135^{c}
			(0.082)		(0.080)	(0.077)	(0.077)
basic Russian skills				0.242	0.267	0.223	0.203
				(0.377)	(0.366)	(0.354)	(0.352)
regular Russian skills				0.172	0.216^{c}	0.238^{b}	0.227^{b}
				(0.115)	(0.113)	(0.110)	(0.110)
advanced Russian skills				0.366^{a}	0.412^{a}	0.345^{b}	$0.356^{\rm b}$
				(0.138)	(0.145)	(0.139)	(0.141)
parental education	no	no	no	no	no	no	yes
occupation fixed effects	no	no	no	no	no	yes	yes
number of observations	9194	9194	9194	9194	9194	9194	9194
log-Likelihood	-12863	-12957	-12955	-12955	-12854	-12471	-12444
pseudo R ²	0.204	0.204	0.204	0.204	0.206	0.236	0.238

Note: robust standard errors within parenthesis in italic. ^c Significant at p < 0.1, ^b significant at p < 0.05, ^a significant at p < 0.01. All the models include controls for years of schooling, potential experience (quadratic), labor market status and a dummy for urban area.

Table 7: heterogeneous returns to foreign language skills — frequency of language use at work

bie 7. heterogeneous returns to foreign language skins –	model 1 model 2 model 2				
	model 1	model 2	model 3		
no English skills	refe	rence cate	gory		
basic English skills - not used at work	0.007	0.010	0.008		
	(0.018)	(0.017)	(0.017)		
basic English skills - used less than once per month	0.104^{c}	0.087	0.080		
	(0.057)	(0.054)	(0.053)		
basic English skills - used at least once per month	0.272^{a}	0.228^{a}	0.219^{a}		
	(0.058)	(0.059)	(0.059)		
basic English skills - used at least once per week	0.090	0.016	0.006		
	(0.062)	(0.061)	(0.061)		
basic English skills - daily used	0.115	0.116	0.104		
	(0.097)	(0.093)	(0.093)		
regular English skills - not used at work	0.154^{a}	0.134^{a}	0.126^{a}		
	(0.029)	(0.028)	(0.028)		
regular English skills - used less than once per month	0.206^{b}	0.164^{b}	0.157^{b}		
	(0.082)	(0.071)	(0.071)		
regular English skills - used at least once per month	0.376^{a}	0.344^{a}	0.330^{a}		
	(0.063)	(0.061)	(0.061)		
regular English skills - used at least once per week	0.228^{a}	0.167^{a}	0.165^{a}		
	(0.063)	(0.060)	(0.060)		
regular English skills - daily used	0.223^{a}	0.172^{a}	0.165^{a}		
	(0.057)	(0.053)	(0.052)		
advanced English skills - not used at work	0.368^{a}	0.335^{a}	0.335^{a}		
	(0.075)	(0.075)	(0.074)		
advanced English skills - used less than once per month	0.382^{a}	0.312^{b}	0.301^{b}		
	(0.120)	(0.124)	(0.125)		
advanced English skills - used at least once per month	0.423^{a}	0.314^{a}	0.301^{b}		
	(0.134)	(0.121)	(0.122)		
advanced English skills - used at least once per week	0.520^{a}	0.453^{a}	0.441^{a}		
	(0.095)	(0.093)	(0.093)		
advanced English skills - daily used	0.502^{a}	0.412^{a}	0.399^{a}		
	(0.050)	(0.050)	(0.050)		
parental education	no	no	yes		
occupation fixed effects	no	yes	yes		
number of observations	9194	9194	9194		
log-Likelihood	-12773	-12398	-12397		
pseudo R ²	0.206	0.236	0.236		

Note: robust standard errors within parenthesis in italic. ^c Significant at p < 0.1, ^b significant at p < 0.05, ^a significant at p < 0.01. All the models include controls for years of schooling, potential experience (quadratic), labor market status and a dummy for urban area.

Table 8: heterogeneous returns to foreign language skills — birth-cohort

	C	ohort 25-	39	С	cohort 40-65			
no English skills	reference category			reference category				
				; !				
basic English skills	0.011	0.013	0.009	$0.068^{\rm b}$	0.048	0.044		
	(0.019)	(0.018)	(0.018)	(0.032)	(0.031)	(0.031)		
regular English skills	0.154^{a}	0.125^{a}	0.117^{a}	0.289^{a}	0.241^{a}	0.228^{a}		
	(0.027)	(0.026)	(0.026)	(0.048)	(0.046)	(0.046)		
advanced English skills	0.373^{a}	0.299^{a}	0.287^{a}	0.705^{a}	0.642^{a}	0.625^{a}		
_	(0.043)	(0.042)	(0.043)	(0.087)	(0.086)	(0.086)		
parental education	no	no	yes	no	no	yes		
occupation fixed effects	no	yes	yes	no	yes	yes		
number of observations	5011	5011	5011	4083	4083	4083		
log-likelihood	-7299	-7046	-7028	-5500	-5346	-5335		
pseudo R ²	0.187	0.230	0.234	0.217	0.243	0.246		

Note: robust standard errors within parenthesis in italic. c Significant at p<0.1, b significant at p<0.05, a significant at p<0.01. All the models include controls for years of schooling, potential experience (quadratic), labor market status and a dummy for urban area.

Cohort 25-39 = individuals aged between 25 and 39. Cohort 40-65 = individuals aged between 40 and 65.

Table 9: heterogeneous returns to foreign language skills — education

	high education			med	medium education			low education		
no English skills	refe	rence cate	egory	refe	reference category			reference category		
basic English skills	0.023	0.036	0.035	0.026 (0.021)	0.022	0.017	0.121 ^b (0.050)	0.083 ^c (0.045)	0.074	
regular English skills	$0.110^{\acute{a}}$	$0.099^{\acute{a}}$	$0.098^{\acute{a}}$	$0.180^{\acute{a}}$	$0.142^{\acute{a}}$	$0.127^{\acute{a}}$	$0.337^{\acute{a}}$	0.264^{a}	$0.239^{\acute{a}}$	
advanced English skills	(0.036) 0.321 ^a (0.047)	(0.036) 0.288 ^a (0.047)	(0.036) 0.292 ^a (0.048)	(0.033) 0.439 ^a (0.075)	(0.031) 0.339^{a} (0.079)	(0.031) 0.317 ^a (0.081)	(0.093) 0.467 ^b (0.189)	(0.091) 0.468 ^a (0.163)	(0.091) 0.447 ^a (0.173)	
parental education	no	no	ves	no	no	ves	no	no	ves	
occupation fixed effects	no	yes	yes	no	yes	yes	no	yes	yes	
number of observations	1488	1488	1488	3211	3211	3211	4495	4495	4495	
log-likelihood	-1494	-1432	-1429	-4775	-4623	-4615	-6493	-6286	-6266	
pseudo R ²	0.084	0.286	0.286	0.113	0.163	0.165	0.113	0.159	0.166	

Note: robust standard errors within parenthesis in italic. ^c Significant at p<0.1, ^b significant at p<0.05, ^a significant at p<0.01. All the models include controls for years of schooling, potential experience (quadratic), labor market status and a dummy for urban area.

High education = *college education or more.*

 $Medium\ education = upper-and-lower\ secondary\ education.$

Low education = primary education or less.

Table 10: heterogeneous returns to foreign language skills — birth-cohort & education

	c	ohort 25-39	9	cohort 40-65			
no English skills	reference category			refe	reference category		
basic English skills	0.117 ^b	0.065	0.048	0.205 ^b	0.161 ^b	0.146 ^c	
regular English skills	(0.055) 0.238 ^b	(0.051) 0.204 ^b	(0.051) 0.171^{c}	(0.084) 0.482 ^a	(0.076) 0.343 ^b	(0.076) 0.316^{b}	
	(0.118)	(0.101)	(0.098)	(0.142)	(0.146)	(0.146)	
advanced English skills	0.712^{a} (0.210)	0.588 ^a (0.168)	0.580^{a} (0.192)	0.232 (0.149)	0.245 (0.211)	0.193 (0.203)	
medium education * basic English skills	-0.056	-0.017	-0.005	-0.157^{c}	-0.138	-0.130	
Medium education * regular English skills	(0.059) -0.034	(0.056) -0.049	(0.056) -0.027	(0.094) -0.167	(0.086) -0.088	(0.086) -0.085	
	(0.124)	(0.107)	(0.104)	(0.158)	(0.159)	(0.159)	
Medium education * advanced English skills	-0.388^{c} (0.225)	-0.371° (0.190)	-0.384^{c} (0.211)	0.672° (0.219)	0.599^{b} (0.266)	0.620 ^b (0.263)	
high education * basic English skills	-0.155^{b}	-0.067	-0.054	-0.108	-0.080	-0.069	
High education * regular English skills	(0.070) -0.173	(0.066) -0.126	(0.067) -0.099	(0.105) -0.231	(0.097) -0.120	(0.097) -0.106	
High education * advanced English skills	(0.127) -0.438 ^b	(0.110) -0.337°	(0.108) -0.341°	(0.160) 0.442 ^b	(0.163) 0.362	(0.163) 0.397 ^c	
rigii education · advanced Eligiisii skiiis	(0.217)	-0.337 (0.177)	(0.199)	(0.191)	(0.239)	(0.231)	
parental education	no	no	yes	no	no	yes	
occupation fixed effects	no	yes	yes	no	yes	yes	
number of observations	5011	5011	5011	4083	4083	4083	
log-likelihood	-7301	-7048	-7023	-5510	-5350	-5335	
pseudo R ²	0.187	0.230	0.234	0.237	0.243	0.246	

Note: robust standard errors within parenthesis in italic. ^c Significant at p<0.1, ^b significant at p<0.05, ^a significant at p<0.01. All the models include controls schooling dummies (high and medium education, low education as reference category), potential experience (quadratic), labor market status and a dummy for urban area. The base-level coefficients for English skills represent the earning return to English skills for low-educated workers (reference category).

 ${\bf Table~12:~heterogeneous~returns~to~foreign~language~skills -- urban/rural~areas} \\$

ibie 121 neterogeneous ret	2. heterogeneous returns to foreign language skins — urban/rurar areas							
	1	urban are	ea	 	rural area			
no English skills	refe	rence cate	egory	reference category				
basic English skills	0.015	0.015	0.014	0.092 ^b	0.077 ^b (0.039)	0.065 ^c (0.039)		
regular English skills	0.198^{a} (0.025)	0.165^{a} (0.024)	0.159^{a} (0.024)	0.188 ^a (0.059)	0.138^{b} (0.058)	0.127^{b} (0.059)		
advanced English skills	$0.470^{\acute{a}}$	0.392^{a}	$0.384^{\acute{a}}$	0.365^{a}	0.294^{6}	0.284^{b}		
	(0.040)	(0.039)	(0.039)	(0.117)	(0.117)	(0.120)		
parental education	no	no	yes	no	no	yes		
occupation fixed effects	no	yes	yes	no	yes	yes		
number of observations	6528	6528	6528	2666	2666	2666		
log-likelihood	-9089	-8829	-8828	-3630	-3530	-3530		
pseudo R ²	0.182	0.220	0.222	0.160	0.192	0.193		

Note: robust standard errors within parenthesis in italic. ^c Significant at p<0.1, ^b significant at p<0.05, ^a significant at p<0.01. All the models include controls for years of schooling, potential experience (quadratic) and labor market status.

Urban area = individuals residing in urban areas.

 $Rural\ area=individuals\ residing\ in\ rural\ areas.$

Table 13: heterogeneous returns to foreign language skills — occupation

	high-skilled white-collars			high-skilled blue-collars			low-ski	lled white	e-collars	low-skilled blue-collars		
no English skills	reference category			reference category			reference category			reference category		
basic English skills	-0.003 (0.027)	-0.008 (0.027)	-0.011 (0.027)	0.063 (0.039)	0.066 ^c (0.038)	0.058 (0.038)	0.058	0.056 (0.035)	0.056 (0.036)	0.023 (0.033)	0.025 (0.032)	0.023 (0.032)
regular English skills	0.175^{a} (0.032)	0.166^{a} (0.032)	0.155^{a} (0.032)	0.150^{b} (0.068)	0.168° (0.068)	0.145^{6} (0.067)	0.118^{6} (0.048)	0.117^{6} (0.048)		0.145^{6} (0.070)	0.145^{6} (0.068)	0.153^{6} (0.068)
advanced English skills	0.380^{a} (0.048)	0.353 ^a (0.048)	0.333^{a} (0.049)	0.635 ^a (0.159)	0.616^{a} (0.149)	0.602 ^a (0.153)	0.333 ^a (0.075)	0.336^{a} (0.078)	,	0.409 ^b (0.163)	0.356^{c} (0.183)	0.388^{b} (0.171)
parental education	no	no	yes	no	no	yes	no	no	yes	no	no	yes
occupation fixed effects	no	yes	yes	no	yes	yes	no	yes	yes	yes	yes	yes
number of observations	2917	2917	2917	2875	2875	2875	1325	1325	1325	2077	2077	2077
log-likelihood	-3606	-3562	-3558	-4062	-4014	-3997	-1922	-1916	-1911	-3002	-2899	-2888
pseudo R ²	0.152	0.157	0.157	0.133	0.146	0.151	0.189	0.195	0.197	0.111	0.161	0.166

Note: robust standard errors within parenthesis in italic. ^c Significant at p < 0.1, ^b significant at p < 0.05, ^a significant at p < 0.01.

All the models include controls for years of schooling, potential experience (quadratic), labor market status and a dummy for urban area.

High-skilled White collar = individuals with 2-digits ISCO88 comprised between 11 and 34.

Low-skilled White collar = individuals with 2-digits ISCO88 comprised between 41 and 52.

High-skilled Blue collar = individuals with 2-digits ISCO88 comprised between 61 and 74.

Low-skilled Blue collar = individuals with 2-digits ISCO88 comprised between 81 and 93.

Table 14: endogenous foreign language skills and earnings

EARNINGS EQUATION	LINEAR (LPM) FIRST STAGE			PROBIT FIRST STAGE				ORPOBIT FIRST STAGE				
proficient in English	0.698	0.501	0.467	0.444	0.711	0.527	0.450	0.416	I I I			
	$(0.089)^a$	$(0.124)^a$	$(0.121)^a$	$(0.126)^a$	$(0.062)^a$	$(0.082)^a$	$(0.086)^a$	$(0.093)^a$	1 ! !			
no English skills					1 1 1				1 ! !	reference	category	
					1 1 1				! ! !			
basic English skills					! ! !				0.117	0.106	0.091	0.091
					! ! !				$(0.031)^a$	$(0.041)^b$	$(0.041)^b$	$(0.042)^b$
regular English skills					! ! !				0.312	0.276	0.238	0.236
1 15 11 121					:				$(0.043)^a$	$(0.059)^a$	$(0.059)^a$	$(0.061)^a$
advanced English skills					!				0.610	0.556	0.482	0.481
ELDOM COL COL					<u> </u>				$(0.062)^a$	$(0.087)^a$	$(0.087)^a$	$(0.090)^a$
FIRST STAGE	, 1	. , . ,			! ! !				: !			
frequency of English use for leisu	re (exclusi				! ! !	c			! ! !	c		
no English or not used		reference category			! ! !	reference	category		reference category			
loss than an as man month	0.157	0.114	0.108	0.101	0.877	0.581	0.569	0.547	1.330	1.024	1.034	1.015
less than once per month	$(0.027)^a$	$(0.026)^a$:		0.568		$(0.065)^a$			
at least once nor month	0.027)	0.020	$(0.026)^a$ 0.183	$(0.026)^a$ 0.173	$(0.106)^a$ 1.168	$(0.119)^a$ 0.810	$(0.121)^a$ 0.819	$(0.122)^a$ 0.802	1.610	$(0.071)^a$ 1.218	$(0.071)^a$ 1.238	$(0.072)^a$ 1.209
at least once per month	$(0.040)^a$	$(0.040)^a$	$(0.040)^a$	$(0.040)^a$	$(0.137)^a$	$(0.157)^a$	$(0.158)^a$	$(0.163)^a$	$(0.098)^a$	$(0.111)^a$	$(0.110)^a$	$(0.112)^a$
at least once per week	0.408	0.326	0.326	0.313	1.558	1.121	1.177	1.145	1.899	1.481	1.509	1.471
at least once per week	$(0.041)^a$	$(0.040)^a$	$(0.040)^a$	$(0.040)^a$	$(0.114)^a$	$(0.132)^a$	$(0.131)^a$	$(0.132)^a$	$(0.092)^a$	$(0.103)^a$	$(0.100)^a$	$(0.101)^a$
daily used	0.394	0.274	0.268	0.259	1.495	0.829	0.873	0.869	1.771	1.207	1.249	1.233
dairy asea	$(0.050)^a$	$(0.050)^a$	$(0.049)^a$	$(0.049)^a$	$(0.161)^a$	$(0.167)^a$	$(0.162)^a$	$(0.163)^a$	$(0.132)^a$	$(0.140)^a$	$(0.140)^a$	$(0.142)^a$
parental education	no	no	no	yes	no	no	no	yes	no	no	no	yes
occupation fixed effects	no	no	yes	yes	no	no	yes	yes	no	no	yes	yes
frequency of English use at work	no	yes	yes	yes	no	yes	yes	yes	no	yes	yes	yes
rho	-0.117	-0.066	-0.069	-0.063	-0.442	-0.296	-0.254	-0.217	-0.139	-0.137	-0.116	-0.121
	$(0.031)^a$	(0.040)	$(0.041)^{c}$	(0.043)	$(0.067)^a$	$(0.098)^a$	$(0.108)^b$	$(0.119)^{c}$	$(0.043)^a$	$(0.057)^b$	$(0.060)^b$	$(0.061)^b$
overidentification test (p-value)	0.061	0.379	0.365	0.409								
significance of excl. restr. (p-value)	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
exogeneity test (p-value)	0.000	0.107	0.093	0.140	0.000	0.005	0.025	0.078	0.001	0.017	0.056	0.051

Note: robust standard errors within parenthesis in italic. ^c Significant at p < 0.1, ^b significant at p < 0.05, ^a significant at p < 0.01.

All the models include controls for years of schooling, potential experience (quadratic), labor market status and a dummy for urban area.

Rho represents the estimated correlation coefficient between the earnings equation's residual and the first stage's residual. The exogeneity test consists in a χ^2 test for the null hypothesis that the estimated rho coefficient is equal to zero.

Table 1A: descriptive statistics

	ALL		NO	TH	AT LI		BEST	
	SAM		NO FL		ONE FL		ENGI	
COMEDOL WARLANDER	mean	s.d	mean	s.d	mean	s.d	mean	s.d
CONTROL VARIABLES	0.26	2.02	<i>(</i> 70	2.00	11.65	2.15	10.16	2.76
years of schooling	8.26	3.82	6.70	3.00	11.65	3.15	12.16	2.76
age	38.99	9.36	39.73	9.44	37.37	8.95	36.22	8.60
potential experience (= years of schooling – age – 6)	24.67	10.77	26.95	10.49	19.75	9.68	18.05	8.93
urban area	0.710	0.45	0.668	0.47	0.803	0.40	0.832	0.37
employee/permanent contract	0.595	0.49	0.541	0.50	0.714	0.45	0.745	0.44
employee/fixed-term contract	0.059	0.24	0.070	0.26	0.035	0.18	0.029	0.17
self-employed	0.285	0.45	0.336	0.47	0.175	0.38	0.151	0.36
employer	0.060	0.24	0.053	0.22	0.076	0.26	0.075	0.26
PARENTAL EDUCATION								
parental education = no education	0.332	0.47	0.403	0.49	0.176	0.38	0.133	0.34
parental education = primary or less	0.588	0.49	0.568	0.50	0.631	0.48	0.639	0.48
parental education = secondary	0.046	0.21	0.020	0.14	0.103	0.30	0.121	0.33
parental education = tertiary	0.034	0.18	0.008	0.09	0.090	0.29	0.107	0.31
OCCUPATION								
high-skilled white collars	0.317	0.47	0.210	0.41	0.552	0.50	0.576	0.49
low-skilled white collars	0.144	0.35	0.131	0.34	0.173	0.38	0.187	0.39
high-skilled blue collars	0.313	0.46	0.392	0.49	0.139	0.35	0.119	0.32
low-skilled blue collars	0.226	0.42	0.267	0.44	0.136	0.34	0.118	0.32
FREQUENCY OF ENGLISH USE AT WORK AN	D FOR L	EISURE	C					
no English skills	0.759	0.43			0.231	0.42		
English not used at work	0.178	0.38			0.568	0.50	0.739	0.44
English used less than once per month at work	0.024	0.15			0.076	0.27	0.099	0.30
English used at least once per month at work	0.013	0.11			0.042	0.20	0.055	0.23
English used at least once per week at work	0.015	0.12			0.048	0.21	0.063	0.24
English daily used at work	0.011	0.10			0.034	0.18	0.044	0.21
English not used for leisure	0.759	0.43			0.231	0.42		
English used less than once per month for leisure	0.158	0.36			0.503	0.50	0.655	0.48
English used at least once per month for leisure	0.017	0.13			0.054	0.23	0.070	0.25
English used at least once per week	0.017	0.13			0.057	0.23	0.075	0.25
English used at least once per week for leisure	0.010	0.13			0.063	0.23	0.073	0.28
English daily used for leisure	0.020				0.003		0.082	
English daily used for leisure	0.028	0.17			0.091	0.29	0.116	0.32

Definition of years of schooling: illiterate = 0 years; literate with no formal education = 2; uncompleted primary education = 3.5; completed primary school = 5; uncompleted middle school = 6.5; completed middle school = 8; uncompleted high school = 9.5; completed high school = 11; uncompleted short college degree = 12; completed short college degree = 13; uncompleted college degree = 14; completed college degree = 15; uncompleted PhD = 17; completed PhD = 19.

Table 2A: foreign language knowledge and earnings among the subsample of English speakers

	French	German	Arabic	Russian	all FLs	all FLs	all FLs
knows French	0.204 ^b				0.180^{b}	0.166 ^b	0.161°
	(0.087)				(0.086)	(0.084)	(0.083)
knows German		0.100^{c}			0.070	0.054	0.049
		(0.053)			(0.054)	(0.048)	(0.048)
knows Arabic			-0.107^{c}		-0.100^{c}	-0.099^{c}	-0.089
			(0.057)		(0.057)	(0.059)	(0.060)
knows Russian				0.164^{b}	0.120^{c}	0.120^{c}	0.112^{c}
				(0.068)	(0.070)	(0.064)	(0.064)
occupation fixed effects	no	no	no	no	no	yes	yes
number of observations	2219	2219	2219	2219	2219	2219	2219
log-Likelihood	-2774	-2775	-2776	-2775	-2660	-2657	-2774
pseudo R ²	0.173	0.173	0.173	0.173	0.174	0.221	0.221

Note: robust standard errors within parenthesis in italic. ^c Significant at p<0.1, ^b significant at p<0.05, ^a significant at p<0.01. All the models include controls for years of schooling, potential experience (quadratic), labor market status and a dummy for urban area. Estimations based on the subsample of individuals who declare English to be the best FL they know. Knowledge of Bulgarian is not considered in this analysis because of the reduced number of observations in this subsample.



2006

CREAP2006-01

Matas, A. (GEAP); Raymond, J.Ll. (GEAP)

"Economic development and changes in car ownership patterns" (Juny 2006)

CREAP2006-02

Trillas, F. (IEB); Montolio, D. (IEB); Duch, N. (IEB)

"Productive efficiency and regulatory reform: The case of Vehicle Inspection Services" (Setembre 2006)

CREAP2006-03

Bel, G. (PPRE-IREA); Fageda, X. (PPRE-IREA)

"Factors explaining local privatization: A meta-regression analysis" (Octubre 2006)

CREAP2006-04

Fernàndez-Villadangos, L. (PPRE-IREA)

"Are two-part tariffs efficient when consumers plan ahead?: An empirical study" (Octubre 2006)

CREAP2006-05

Artís, M. (AQR-IREA); Ramos, R. (AQR-IREA); Suriñach, J. (AQR-IREA)

"Job losses, outsourcing and relocation: Empirical evidence using microdata" (Octubre 2006)

CREAP2006-06

Alcañiz, M. (RISC-IREA); Costa, A.; Guillén, M. (RISC-IREA); Luna, C.; Rovira, C.

"Calculation of the variance in surveys of the economic climate" (Novembre 2006)

CREAP2006-07

Albalate, D. (PPRE-IREA)

"Lowering blood alcohol content levels to save lives: The European Experience" (Desembre 2006)

CREAP2006-08

Garrido, A. (IEB); Arqué, P. (IEB)

"The choice of banking firm: Are the interest rate a significant criteria?" (Desembre 2006)

CREAP2006-09

 $\textbf{Segarra, A.} \ (\textbf{GRIT}) \textbf{; Teruel-Carrizosa, M.} \ (\textbf{GRIT})$

"Productivity growth and competition in spanish manufacturing firms:

What has happened in recent years?"

(Desembre 2006)

CREAP2006-10

Andonova, V.; Díaz-Serrano, Luis. (CREB)

"Political institutions and the development of telecommunications" (Desembre 2006)

CREAP2006-11

Raymond, J.L.(GEAP); Roig, J.L.. (GEAP)

"Capital humano: un análisis comparativo Catalunya-España" (Desembre 2006)

CREAP2006-12

Rodríguez, M.(CREB); Stoyanova, A. (CREB)

"Changes in the demand for private medical insurance following a shift in tax incentives" (Desembre 2006)



CREAP2006-13

Royuela, V. (AQR-IREA); Lambiri, D.; Biagi, B.

"Economía urbana y calidad de vida. Una revisión del estado del conocimiento en España" (Desembre 2006)

CREAP2006-14

Camarero, M.; Carrion-i-Silvestre, J.LL. (AQR-IREA).; Tamarit, C.

"New evidence of the real interest rate parity for OECD countries using panel unit root tests with breaks" (Desembre 2006)

CREAP2006-15

Karanassou, M.; Sala, H. (GEAP).; Snower, D. J.

"The macroeconomics of the labor market: Three fundamental views" (Desembre 2006)

2007

XREAP2007-01

Castany, L (AQR-IREA); López-Bazo, E. (AQR-IREA).; Moreno, R. (AQR-IREA)

"Decomposing differences in total factor productivity across firm size" (Març 2007)

XREAP2007-02

Raymond, J. Ll. (GEAP); Roig, J. Ll. (GEAP)

"Una propuesta de evaluación de las externalidades de capital humano en la empresa" (Abril 2007)

XREAP2007-03

Durán, J. M. (IEB); Esteller, A. (IEB)

"An empirical analysis of wealth taxation: Equity vs. Tax compliance" (Juny 2007)

XREAP2007-04

Matas, A. (GEAP); Raymond, J.Ll. (GEAP)

"Cross-section data, disequilibrium situations and estimated coefficients: evidence from car ownership demand" (Juny 2007)

XREAP2007-05

Jofre-Montseny, J. (IEB); Solé-Ollé, A. (IEB)

"Tax differentials and agglomeration economies in intraregional firm location" (Juny 2007)

XREAP2007-06

Álvarez-Albelo, C. (CREB); Hernández-Martín, R.

"Explaining high economic growth in small tourism countries with a dynamic general equilibrium model" (Juliol 2007)

XREAP2007-07

Duch, N. (IEB); Montolio, D. (IEB); Mediavilla, M.

"Evaluating the impact of public subsidies on a firm's performance: a quasi-experimental approach" (Juliol 2007)

XREAP2007-08

Segarra-Blasco, A. (GRIT)

"Innovation sources and productivity: a quantile regression analysis" (Octubre 2007)

XREAP2007-09

Albalate, D. (PPRE-IREA)

"Shifting death to their Alternatives: The case of Toll Motorways" (Octubre 2007)

XREAP2007-10

Segarra-Blasco, A. (GRIT); Garcia-Quevedo, J. (IEB); Teruel-Carrizosa, M. (GRIT)

"Barriers to innovation and public policy in catalonia" (Novembre 2007)



XREAP2007-11

Bel, G. (PPRE-IREA); Foote, J.

"Comparison of recent toll road concession transactions in the United States and France" (Novembre 2007)

XREAP2007-12

Segarra-Blasco, A. (GRIT);

"Innovation, R&D spillovers and productivity: the role of knowledge-intensive services" (Novembre 2007)

XREAP2007-13

Bermúdez Morata, Ll. (RFA-IREA); Guillén Estany, M. (RFA-IREA), Solé Auró, A. (RFA-IREA)

"Impacto de la inmigración sobre la esperanza de vida en salud y en discapacidad de la población española" (Novembre 2007)

XREAP2007-14

Calaeys, P. (AQR-IREA); Ramos, R. (AQR-IREA), Suriñach, J. (AQR-IREA)

"Fiscal sustainability across government tiers" (Desembre 2007)

XREAP2007-15

Sánchez Hugalbe, A. (IEB)

"Influencia de la inmigración en la elección escolar" (Desembre 2007)

2008

XREAP2008-01

Durán Weitkamp, C. (GRIT); Martín Bofarull, M. (GRIT); Pablo Martí, F.

"Economic effects of road accessibility in the Pyrenees: User perspective" (Gener 2008)

XREAP2008-02

Díaz-Serrano, L.; Stoyanova, A. P. (CREB)

"The Causal Relationship between Individual's Choice Behavior and Self-Reported Satisfaction: the Case of Residential Mobility in the EU" (Març 2008)

XREAP2008-03

Matas, A. (GEAP); Raymond, J. L. (GEAP); Roig, J. L. (GEAP)

"Car ownership and access to jobs in Spain" (Abril 2008)

XREAP2008-04

Bel, G. (PPRE-IREA); Fageda, X. (PPRE-IREA)

"Privatization and competition in the delivery of local services: An empirical examination of the dual market hypothesis" (Abril 2008)

XREAP2008-05

Matas, A. (GEAP); Raymond, J. L. (GEAP); Roig, J. L. (GEAP)

"Job accessibility and employment probability" (Maig 2008)

XREAP2008-06

Basher, S. A.; Carrión, J. Ll. (AQR-IREA)

Deconstructing Shocks and Persistence in OECD Real Exchange Rates (Juny 2008)

XREAP2008-07

Sanromá, E. (IEB); Ramos, R. (AQR-IREA); Simón, H.

Portabilidad del capital humano y asimilación de los inmigrantes. Evidencia para España (Juliol 2008)

XREAP2008-08

Basher, S. A.; Carrión, J. Ll. (AQR-IREA)

Price level convergence, purchasing power parity and multiple structural breaks: An application to US cities (Juliol 2008)

XREAP2008-09

Bermúdez, Ll. (RFA-IREA)

A priori ratemaking using bivariate poisson regression models (Juliol 2008)

XREAP2008-10

Solé-Ollé, A. (IEB), Hortas Rico, M. (IEB)

Does urban sprawl increase the costs of providing local public services? Evidence from Spanish municipalities (Novembre 2008)

XREAP2008-11

Teruel-Carrizosa, M. (GRIT), Segarra-Blasco, A. (GRIT)

Immigration and Firm Growth: Evidence from Spanish cities (Novembre 2008)

XREAP2008-12

Duch-Brown, N. (IEB), García-Quevedo, J. (IEB), Montolio, D. (IEB)

Assessing the assignation of public subsidies: Do the experts choose the most efficient R&D projects? (Novembre 2008)

XREAP2008-13

Bilotkach, V., Fageda, X. (PPRE-IREA), Flores-Fillol, R.

Scheduled service versus personal transportation: the role of distance (Desembre 2008)

XREAP2008-14

Albalate, D. (PPRE-IREA), Gel, G. (PPRE-IREA)

Tourism and urban transport: Holding demand pressure under supply constraints (Desembre 2008)

2009

XREAP2009-01

Calonge, S. (CREB); Tejada, O.

"A theoretical and practical study on linear reforms of dual taxes" (Febrer 2009)

Albalate, D. (PPRE-IREA); Fernández-Villadangos, L. (PPRE-IREA)

"Exploring Determinants of Urban Motorcycle Accident Severity: The Case of Barcelona" (Marc 2009)

XREAP2009-03

Borrell, J. R. (PPRE-IREA): Fernández-Villadangos, L. (PPRE-IREA)

"Assessing excess profits from different entry regulations" (Abril 2009)

XREAP2009-04

Sanromá, E. (IEB); Ramos, R. (AQR-IREA), Simon, H.

"Los salarios de los inmigrantes en el mercado de trabajo español. ¿Importa el origen del capital humano?" (Abril 2009)

XREAP2009-05

Jiménez, J. L.; Perdiguero, J. (PPRE-IREA)

"(No)competition in the Spanish retailing gasoline market: a variance filter approach" (Maig 2009)

XREAP2009-06

Álvarez-Albelo, C. D. (CREB), Manresa, A. (CREB), Pigem-Vigo, M. (CREB)

"International trade as the sole engine of growth for an economy" (Juny 2009)

XREAP2009-07

Callejón, M. (PPRE-IREA), Ortún V, M.

"The Black Box of Business Dynamics"

(Setembre 2009)



XREAP2009-08

Lucena, A. (CREB)

"The antecedents and innovation consequences of organizational search: empirical evidence for Spain" (Octubre 2009)

XREAP2009-09

Domènech Campmajó, L. (PPRE-IREA)

"Competition between TV Platforms" (Octubre 2009)

XREAP2009-10

Solé-Auró, A. (RFA-IREA), Guillén, M. (RFA-IREA), Crimmins, E. M.

"Health care utilization among immigrants and native-born populations in 11 European countries. Results from the Survey of Health, Ageing and Retirement in Europe"

(Octubre 2009)

XREAP2009-11

Segarra, A. (GRIT), Teruel, M. (GRIT)

"Small firms, growth and financial constraints" (Octubre 2009)

XREAP2009-12

Matas, A. (GEAP), Raymond, J.Ll. (GEAP), Ruiz, A. (GEAP)

"Traffic forecasts under uncertainty and capacity constraints" (Novembre 2009)

XREAP2009-13

Sole-Ollé, A. (IEB)

"Inter-regional redistribution through infrastructure investment: tactical or programmatic?" (Novembre 2009)

XREAP2009-14

Del Barrio-Castro, T., García-Quevedo, J. (IEB)

"The determinants of university patenting: Do incentives matter?" (Novembre 2009)

XREAP2009-15

Ramos, R. (AQR-IREA), Suriñach, J. (AQR-IREA), Artís, M. (AQR-IREA)

"Human capital spillovers, productivity and regional convergence in Spain" (Novembre 2009)

XREAP2009-16

Álvarez-Albelo, C. D. (CREB), Hernández-Martín, R.

"The commons and anti-commons problems in the tourism economy" (Desembre 2009)

2010

XREAP2010-01

García-López, M. A. (GEAP)

"The Accessibility City. When Transport Infrastructure Matters in Urban Spatial Structure" (Febrer 2010)

XREAP2010-02

García-Quevedo, J. (IEB), Mas-Verdú, F. (IEB), Polo-Otero, J. (IEB)

"Which firms want PhDs? The effect of the university-industry relationship on the PhD labour market" (Març 2010)

XREAP2010-03

Pitt. D., Guillén, M. (RFA-IREA)

"An introduction to parametric and non-parametric models for bivariate positive insurance claim severity distributions" (Març 2010)



XREAP2010-04

Bermúdez, Ll. (RFA-IREA), Karlis, D.

"Modelling dependence in a ratemaking procedure with multivariate Poisson regression models" (Abril 2010)

XREAP2010-05

Di Paolo, A. (IEB)

"Parental education and family characteristics: educational opportunities across cohorts in Italy and Spain" (Maig 2010)

XREAP2010-06

Simón, H. (IEB), Ramos, R. (AQR-IREA), Sanromá, E. (IEB)

"Movilidad ocupacional de los inmigrantes en una economía de bajas cualificaciones. El caso de España" (Juny 2010)

XREAP2010-07

Di Paolo, A. (GEAP & IEB), Raymond, J. Ll. (GEAP & IEB)

"Language knowledge and earnings in Catalonia" (Juliol 2010)

XREAP2010-08

Bolancé, C. (RFA-IREA), Alemany, R. (RFA-IREA), Guillén, M. (RFA-IREA)

"Prediction of the economic cost of individual long-term care in the Spanish population" (Setembre 2010)

XREAP2010-09

Di Paolo, A. (GEAP & IEB)

"Knowledge of catalan, public/private sector choice and earnings: Evidence from a double sample selection model" (Setembre 2010)

XREAP2010-10

Coad, A., Segarra, A. (GRIT), Teruel, M. (GRIT)

"Like milk or wine: Does firm performance improve with age?" (Setembre 2010)

XREAP2010-11

Di Paolo, A. (GEAP & IEB), Raymond, J. Ll. (GEAP & IEB), Calero, J. (IEB)

"Exploring educational mobility in Europe" (Octubre 2010)

XREAP2010-12

Borrell, A. (GiM-IREA), Fernández-Villadangos, L. (GiM-IREA)

"Clustering or scattering: the underlying reason for regulating distance among retail outlets" (Desembre 2010)

XREAP2010-13

Di Paolo, A. (GEAP & IEB)

"School composition effects in Spain" (Desembre 2010)

XREAP2010-14

Fageda, X. (GiM-IREA), Flores-Fillol, R.

"Technology, Business Models and Network Structure in the Airline Industry" (Desembre 2010)

XREAP2010-15

Albalate, D. (GiM-IREA), Bel, G. (GiM-IREA), Fageda, X. (GiM-IREA)

"Is it Redistribution or Centralization? On the Determinants of Government Investment in Infrastructure" (Desembre 2010)

XREAP2010-16

Oppedisano, V., Turati, G.

"What are the causes of educational inequalities and of their evolution over time in Europe? Evidence from PISA" (Desembre 2010)

69

SÈRIE DE DOCUMENTS DE TREBALL DE LA XREAP

XREAP2010-17

Canova, L., Vaglio, A.

"Why do educated mothers matter? A model of parental help" (Desembre 2010)

2011

XREAP2011-01

Fageda, X. (GiM-IREA), Perdiguero, J. (GiM-IREA)

"An empirical analysis of a merger between a network and low-cost airlines" (Maig 2011)

XREAP2011-02

Moreno-Torres, I. (ACCO, CRES & GiM-IREA)

"What if there was a stronger pharmaceutical price competition in Spain? When regulation has a similar effect to collusion" (Maig 2011)

XREAP2011-03

Miguélez, E. (AQR-IREA); Gómez-Miguélez, I.

"Singling out individual inventors from patent data" (Maig 2011)

XREAP2011-04

Moreno-Torres, I. (ACCO, CRES & GiM-IREA)

"Generic drugs in Spain: price competition vs. moral hazard" (Maig 2011)

XREAP2011-05

Nieto, S. (AQR-IREA), Ramos, R. (AQR-IREA)

"¿Afecta la sobreeducación de los padres al rendimiento académico de sus hijos?" (Maig 2011)

XREAP2011-06

Pitt, D., Guillén, M. (RFA-IREA), Bolancé, C. (RFA-IREA)

"Estimation of Parametric and Nonparametric Models for Univariate Claim Severity Distributions - an approach using R" (Juny 2011)

XREAP2011-07

Guillén, M. (RFA-IREA), Comas-Herrera, A.

"How much risk is mitigated by LTC Insurance? A case study of the public system in Spain" (Juny 2011)

XREAP2011-08

Ayuso, M. (RFA-IREA), Guillén, M. (RFA-IREA), Bolancé, C. (RFA-IREA)

"Loss risk through fraud in car insurance" (Juny 2011)

XREAP2011-09

Duch-Brown, N. (IEB), García-Quevedo, J. (IEB), Montolio, D. (IEB)

"The link between public support and private R&D effort: What is the optimal subsidy?" (Juny 2011)

XREAP2011-10

Bermúdez, Ll. (RFA-IREA), Karlis, D.

"Mixture of bivariate Poisson regression models with an application to insurance" (Juliol 2011)

XREAP2011-11

Varela-Irimia, X-L. (GRIT)

"Age effects, unobserved characteristics and hedonic price indexes: The Spanish car market in the 1990s" (Agost 2011)

XREAP2011-12

Bermúdez, Ll. (RFA-IREA), Ferri, A. (RFA-IREA), Guillén, M. (RFA-IREA)

"A correlation sensitivity analysis of non-life underwriting risk in solvency capital requirement estimation" (Setembre 2011)



XREAP2011-13

Guillén, M. (RFA-IREA), Pérez-Marín, A. (RFA-IREA), Alcañiz, M. (RFA-IREA)

"A logistic regression approach to estimating customer profit loss due to lapses in insurance" (Octubre 2011)

XREAP2011-14

Jiménez, J. L., Perdiguero, J. (GiM-IREA), García, C.

"Evaluation of subsidies programs to sell green cars: Impact on prices, quantities and efficiency" (Octubre 2011)

XREAP2011-15

Arespa, M. (CREB)

"A New Open Economy Macroeconomic Model with Endogenous Portfolio Diversification and Firms Entry" (Octubre 2011)

XREAP2011-16

Matas, A. (GEAP), Raymond, J. L. (GEAP), Roig, J.L. (GEAP)

"The impact of agglomeration effects and accessibility on wages" (Novembre 2011)

XREAP2011-17

Segarra, A. (GRIT)

"R&D cooperation between Spanish firms and scientific partners: what is the role of tertiary education?" (Novembre 2011)

XREAP2011-18

García-Pérez, J. I.; Hidalgo-Hidalgo, M.; Robles-Zurita, J. A.

"Does grade retention affect achievement? Some evidence from PISA" (Novembre 2011)

XREAP2011-19

Arespa, M. (CREB)

"Macroeconomics of extensive margins: a simple model" (Novembre 2011)

XREAP2011-20

García-Quevedo, J. (IEB), Pellegrino, G. (IEB), Vivarelli, M.

"The determinants of YICs' R&D activity" (Desembre 2011)

XREAP2011-21

González-Val, R. (IEB), Olmo, J.

"Growth in a Cross-Section of Cities: Location, Increasing Returns or Random Growth?" (Desembre 2011)

XREAP2011-22

Gombau, V. (GRIT), Segarra, A. (GRIT)

"The Innovation and Imitation Dichotomy in Spanish firms: do absorptive capacity and the technological frontier matter?" (Desembre 2011)

2012

XREAP2012-01

Borrell, J. R. (GiM-IREA), Jiménez, J. L., García, C.

"Evaluating Antitrust Leniency Programs" (Gener 2012)

XREAP2012-02

Ferri, A. (RFA-IREA), Guillén, M. (RFA-IREA), Bermúdez, Ll. (RFA-IREA)

"Solvency capital estimation and risk measures" (Gener 2012)

XREAP2012-03

Ferri, A. (RFA-IREA), Bermúdez, Ll. (RFA-IREA), Guillén, M. (RFA-IREA)

"How to use the standard model with own data" (Febrer 2012)



XREAP2012-04

Perdiguero, J. (GiM-IREA), Borrell, J.R. (GiM-IREA)

"Driving competition in local gasoline markets" (Marc 2012)

XREAP2012-05

D'Amico, G., Guillen, M. (RFA-IREA), Manca, R.

"Discrete time Non-homogeneous Semi-Markov Processes applied to Models for Disability Insurance" (Març 2012)

XREAP2012-06

Bové-Sans, M. A. (GRIT), Laguado-Ramírez, R.

"Quantitative analysis of image factors in a cultural heritage tourist destination" (Abril 2012)

XREAP2012-07

Tello, C. (AQR-IREA), Ramos, R. (AQR-IREA), Artís, M. (AQR-IREA)

"Changes in wage structure in Mexico going beyond the mean: An analysis of differences in distribution, 1987-2008" (Maig 2012)

XREAP2012-08

Jofre-Monseny, J. (IEB), Marín-López, R. (IEB), Viladecans-Marsal, E. (IEB)

"What underlies localization and urbanization economies? Evidence from the location of new firms" (Maig 2012)

XREAP2012-09

Muñiz, I. (GEAP), Calatayud, D., Dobaño, R.

"Los límites de la compacidad urbana como instrumento a favor de la sostenibilidad. La hipótesis de la compensación en Barcelona medida a través de la huella ecológica de la movilidad y la vivienda" (Maig 2012)

XREAP2012-10

Arqué-Castells, P. (GEAP), Mohnen, P.

"Sunk costs, extensive R&D subsidies and permanent inducement effects" (Maig 2012)

XREAP2012-11

Boj, E. (CREB), Delicado, P., Fortiana, J., Esteve, A., Caballé, A.

"Local Distance-Based Generalized Linear Models using the dbstats package for R" (Maig 2012)

XREAP2012-12

Rovuela, V. (AOR-IREA)

"What about people in European Regional Science?" (Maig 2012)

XREAP2012-13

Osorio A. M. (RFA-IREA), Bolancé, C. (RFA-IREA), Madise, N.

"Intermediary and structural determinants of early childhood health in Colombia: exploring the role of communities" (Juny 2012)

XREAP2012-14

Miguelez. E. (AQR-IREA), Moreno, R. (AQR-IREA)

"Do labour mobility and networks foster geographical knowledge diffusion? The case of European regions" (Juliol 2012)

XREAP2012-15

Teixidó-Figueras, J. (GRIT), Duró, J. A. (GRIT)

"Ecological Footprint Inequality: A methodological review and some results" (Setembre 2012)

XREAP2012-16

Varela-Irimia, X-L. (GRIT)

"Profitability, uncertainty and multi-product firm product proliferation: The Spanish car industry" (Setembre 2012)



XREAP2012-17

Duró, J. A. (GRIT), Teixidó-Figueras, J. (GRIT)

"Ecological Footprint Inequality across countries: the role of environment intensity, income and interaction effects" (Octubre 2012)

XREAP2012-18

Manresa, A. (CREB), Sancho, F.

"Leontief versus Ghosh: two faces of the same coin" (Octubre 2012)

XREAP2012-19

Alemany, R. (RFA-IREA), Bolancé, C. (RFA-IREA), Guillén, M. (RFA-IREA)

"Nonparametric estimation of Value-at-Risk" (Octubre 2012)

XREAP2012-20

Herrera-Idárraga, P. (AQR-IREA), López-Bazo, E. (AQR-IREA), Motellón, E. (AQR-IREA)

"Informality and overeducation in the labor market of a developing country" (Novembre 2012)

XREAP2012-21

Di Paolo, A. (AQR-IREA)

"(Endogenous) occupational choices and job satisfaction among recent PhD recipients: evidence from Catalonia" (Desembre 2012)

2013

XREAP2013-01

Segarra, A. (GRIT), García-Quevedo, J. (IEB), Teruel, M. (GRIT)

"Financial constraints and the failure of innovation projects" (Març 2013)

XREAP2013-02

Osorio, A. M. (RFA-IREA), Bolancé, C. (RFA-IREA), Madise, N., Rathmann, K.

"Social Determinants of Child Health in Colombia: Can Community Education Moderate the Effect of Family Characteristics?" (Març 2013)

XREAP2013-03

Teixidó-Figueras, J. (GRIT), Duró, J. A. (GRIT)

"The building blocks of international ecological footprint inequality: a regression-based decomposition" (Abril 2013)

XREAP2013-04

Salcedo-Sanz, S., Carro-Calvo, L., Claramunt, M. (CREB), Castañer, A. (CREB), Marmol, M. (CREB)

"An Analysis of Black-box Optimization Problems in Reinsurance: Evolutionary-based Approaches" (Maig 2013)

XREAP2013-05

Alcañiz, M. (RFA), Guillén, M. (RFA), Sánchez-Moscona, D. (RFA), Santolino, M. (RFA), Llatje, O., Ramon, Ll.

"Prevalence of alcohol-impaired drivers based on random breath tests in a roadside survey" (Juliol 2013)

XREAP2013-06

Matas, A. (GEAP & IEB), Raymond, J. Ll. (GEAP & IEB), Roig, J. L. (GEAP)

"How market access shapes human capital investment in a peripheral country" (Octubre 2013)

XREAP2013-07

Di Paolo, A. (AQR-IREA), Tansel, A.

"Returns to Foreign Language Skills in a Developing Country: The Case of Turkey" (Novembre 2013)



xarxa.xreap@gmail.com